

Early lead exposure and outcomes in adulthood

Hans Grönqvist J Peter Nilsson Per-Olof Robling

WORKING PAPER 2017:4

The Institute for Evaluation of Labour Market and Education Policy (IFAU) is a research institute under the Swedish Ministry of Employment, situated in Uppsala. IFAU's objective is to promote, support and carry out scientific evaluations. The assignment includes: the effects of labour market and educational policies, studies of the functioning of the labour market and the labour market effects of social insurance policies. IFAU shall also disseminate its results so that they become accessible to different interested parties in Sweden and abroad.

Papers published in the Working Paper Series should, according to the IFAU policy, have been discussed at seminars held at IFAU and at least one other academic forum, and have been read by one external and one internal referee. They need not, however, have undergone the standard scrutiny for publication in a scientific journal. The purpose of the Working Paper Series is to provide a factual basis for public policy and the public policy discussion.

More information about IFAU and the institute's publications can be found on the website www.ifau.se

ISSN 1651-1166

Early lead exposure and outcomes in adulthood¹

by

Hans Grönqvist², J Peter Nilsson³, and Per-Olof Robling⁴

May 15, 2017

Abstract

We exploit the phase-out of leaded gasoline to isolate the impact of early childhood lead exposure on outcomes in adulthood. By combining administrative data on school performance, high school graduation, crime, earnings, and cognitive and non-cognitive skills with a novel measure of lead exposure, we follow 800,000 children from birth into adulthood. We find that reduced lead exposure improves the adult outcomes, particularly among boys. Below certain thresholds, the relationship becomes much weaker. Non-cognitive traits (externalizing behavior, conscientiousness, and neuroticism) follow a similar non-linear dose-response pattern and seem to be the key mediators between early lead exposure and adult outcomes.

Keywords: Environmental policy, human capital; crime, non-cognitive skills. JEL-codes: I18, K42, Q53.

¹ This paper is a merged and substantially revised version of two previous working papers: Nilsson (2009) and Grönqvist, Nilsson and Robling (2014). We are grateful to Anders Björklund, Janet Currie, Per-Anders Edin, Marcus Eliason, Matz Dahlberg, Lena Hensvik, Markus Jäntti, Per Johansson, Patrik Hesselius, Katharina Janke, Per Pettersson-Lidbom, Susan Niknami, Matti Sarvimäki, Staffan Skerfving, David Slusky, David Strömberg, Ulf Strömberg, Germund Tyler and seminar and conference participants at Bristol University, Columbia University, the Econometric Society European Winter Meetings, the Nordic Network in Empirical Labor Economics (Bergen), Princeton University, Stanford University, Uppsala University, SOFI Stockholm University, the Institute for Fiscal Studies (London), Universidad Carlos III de Madrid and the CReAM workshop in Nizza Monferato. Björn Öckert kindly shared data with us, as did Olle Nordell at Landskrona municipality and Staffan Skerfving at Lund University. Part of this research was completed while the authors visited the Center for Health and Wellbeing at Princeton University (Grönqvist and Nilsson), the Department of Economics at UC Berkeley (Robling) and SIEPR Stanford University (Nilsson).

² Department of Economics, Uppsala University, IFAU, UCLS; hans.gronqvist@nek.uu.se

³ Institute for International Economic Studies(IIES), Stockholm University, IFAU, UCLS; peter.nilsson@iies.su.se

⁴ Institute for Social Research (SOFI), Stockholm University; *per-olof.robling@sofi.su.se*

Table of contents

1	Introduction	3
2	Conceptual framework	9
3 3.1 3.2	Background Measuring lead exposure Micro data	11 11 15
4 4.1	Results Baseline results and robustness	17
4.2 4.3 4.4 4.5	Isolating the effects of the 1980/81 reform The mediating role of cognitive and non-cognitive skills Effects on labor market outcomes	23 28 32 33
5	Discussion and economic significance	35
6	Concluding remarks	37
Referen	nces	
Append	lix A.I.: Figures and tables	47
Append	lix A.II.: Sibling fixed effects	61
Append	dix B: The relationship between lead exposure and blood lead among children	ren .63
Append	lix C: Effects on earnings	67

1 Introduction

Lead (Pb) is extremely toxic and high levels of exposure are well documented to cause adverse and often irreversible health effects and even death. Today, exposure causing clinical symptoms is relatively rare, primarily due to policies first limiting and finally banning lead from gasoline. However, the general population is still exposed to lower diffuse doses of lead from a variety of sources including food, air, toys, contaminated soils, lead-based paint, and water.⁵ Recently, in Flint, Michigan, exposure due to deteriorating lead water pipes was made salient when an estimated 6,000 to 12,000 children experienced increased lead exposure, and the share of children aged below 6 that tested positive for elevated blood lead levels (EBLL) more than doubled (Hannah-Attisha et al. 2016). Globally, EBLL is still common: WHO estimates that in 49 percent of all children the blood-lead (B-Pb) level exceeds five micrograms per deciliter blood (5 μ g/dL), i.e., the reference value for elevated blood lead established by the Centers for Disease Control and Prevention (CDC) in 2012.

Lead exposure is believed to be particularly harmful to infants and toddlers because of their rapidly developing central nervous system, increased exposure due to hand-tomouth activities and, conditional on exposure, a higher absorption rate (Etzel 2003). But, understanding whether, and at what level of exposure, lead is negative for long-run human capital outcomes has proven challenging for several reasons. One problem is data availability. The ideal dataset would be a long and representative panel with measures of lead exposure in early childhood, relevant adult outcomes such as highschool completion, criminal behavior, and labor market outcomes, as well as direct measures of the mediating inputs (cognitive and non-cognitive skills) in the production of human capital leading up to outcomes in adulthood. In most countries, however, the required data does not exist. Another major challenge is the threat to identification from correlated unobservables. While it is known that children experiencing high levels of lead exposure have lower cognitive and non-cognitive skills and worse adult outcomes, these correlations could be driven by unobserved characteristics as opposed to early lead exposure.⁶

⁵ For example, the Department of Housing and Urban Development estimates that 38 million homes constructed before 1978 still contain lead based paint.

⁶ Given that lead levels are not randomly distributed across neighborhoods, any study estimating the effects of lead exposure needs to address the concern of confounding factors. For example, parents with higher incomes or preferences for cleaner air are likely to sort into areas with better air quality. Failing to account for residential sorting

Due to these challenges, evidence on the causal effects of low levels of lead exposure on adult outcomes is scarce. In the 1970s and 1980s, several cross-sectional studies documented strong associations between lead exposure and cognitive and behavioral outcomes among children without clinical symptoms. But, by today's standards, even these children experienced extreme levels of exposure (e.g. Landrigan et al. 1975, Needleman et al. 1979).⁷ More recent epidemiological studies have detected cognitive and behavioral deficits at ever lower levels of exposure (e.g. Canfield et al. 2003; Banks et al. 1997). The evidence on effects of low lead exposure is nonetheless still debated (Lanphear et al. 2000). Not only is it unclear at what level damage actually occurs, but the results have been questioned for the use of small and unrepresentative samples,⁸ measurement error in blood-lead and outcomes, a focus on short-run effects, and the failure to properly control for confounding factors (e.g. Ernhart 1995; Kaufman 2001a, 2001b; Heben 2001; Needleman and Bellinger 2001). Bellinger (2004) suggests that developing and controlling for "more differentiated measures [...] of social class", consistency of findings from multiple cross-sectional studies, and taking the evidence from animal experiments into account, could be valuable for drawing causal inference. Arguably, each of these solutions has limitations.⁹

Our study overcomes both the data and the identification issues by approaching this question from a different angle. First, by merging data on local air lead levels with comprehensive population micro data, we are able to measure lead exposure during the first three years of life for about 800,000 children and document their outcomes up to two decades later. The data include multiple markers of development and behavior: compulsory school performance, educational attainment, criminal convictions as well as direct measures of cognitive and non-cognitive skills in adulthood. We use a novel measure of lead exposure based on moss (bryophyte) samples collected all across

of this kind can result in an upwardly biased estimate of the effect of lead exposure on children's subsequent outcomes. However, pollution also tends to be higher in densely populated areas and metropolitan areas attract highly educated parents with more resources, have better access to good child care, health care etc. which could understate the role of lead exposure if not properly taken into account.

 $^{^7}$ In Landrigan et al. (1975), mean B-Pb was $48\mu g/dl$ in the "high exposure" and 27 $\mu g/dl$ in the "low exposure" group.

⁸ Needleman et al. (1996) study 301 first graders in Pittsburgh, Pennsylvania. Dietrich et al. (2001) examine 195 inner city adolescents from Cincinnati, Ohio. The study by Wright et al. (2008) involves 376 children also from Cincinnati, Ohio.

⁹ It is not at all clear that relying on consistent results from multiple cross-sectional studies solves the identification problem since similar confounders are likely to be prevalent in most settings, nor that adding more and more observables will be likely to satisfy the conditional independence assumption. Finally, while evidence from animal studies holds information on the toxicity of lead, it is, for example, unclear how such findings map back to costbenefit analyses of particular policy interventions with respect to effects of human exposure on adult outcomes.

Sweden by the Environmental Protection Agency. Mosses are particularly useful as air pollution biomonitors since they lack roots and therefore solely absorb air-lead depositions.¹⁰ Blood-lead, which is the most commonly used measure of lead exposure in the literature, provides measures of current exposure (half-life of 2 to 3 months), while the moss samples provide a measure of long-term exposure (average over three preceding years). However, in a smaller sample, we do verify the link between changes in local moss lead levels and changes in individual blood.

Second, we address the threats to identification by focusing on children born from the early 1970s until the mid-1980s. The reason is that during the 1970s, Sweden initiated a phase-out of leaded gasoline in order to protect the environment and public health.¹¹ In Sweden, the main reduction in gasoline-lead levels occurred between 1973 and 1981 when the maximum allowed lead level per liter of gasoline dropped by 80%. Since gasoline lead was the main source of lead exposure in the general population,¹² there was a sharp decrease in children's blood lead levels from the 1970s until the mid-1990s, when leaded gasoline was banned. Due to large differences in the initial lead levels, the phase-out of leaded gasoline induced a substantial variation across localities in the reduction of lead exposure. Our main estimation strategy exploits the differential within-neighborhood changes in early childhood lead exposure that occurred for the cohorts born between 1972 and 1984. By exploiting these sharp changes in exposure across birth cohorts within neighborhoods, we are not only able to account for observable background characteristics, but importantly also for neighborhood-specific unobservables. In addition, exploiting the timing of the changes in maximum allowed lead content following the major reforms in 1980/81, we trace-out how the reform effect evolves across cohorts.

Our study provides comprehensive evidence on the effects of low-levels of early lead exposure on adult outcomes. Our baseline linear specification shows that reducing early lead exposure improves compulsory school grade point averages, increases the likelihood of high school completion, and reduces the risk of being convicted before the age of 25. We also document stronger effects for boys, and for children from low

¹⁰ The use of moss as biomonitors for heavy metal air pollution is well established today and used in many countries.
¹¹ Other parts of the developing world lagged far behind. For example, a continent-wide ban on leaded gasoline in

Africa was not reached until 2006 (with the exception of Algeria).

 $^{^{12}}$ 80% of the air lead levels in the late 1980s, i.e. after the bulk of the reforms, were still due to traffic (MOENR 1994). Lead-based paint was banned in the 1920s in Sweden (in the end of the 1970s in the US).

income households. The results are robust to a wide range of specification checks, such as controlling for exposure to other pollutants and local economic conditions as well as testing for endogenous parental sorting.

The identification of thresholds, below which the effects of reductions in exposure have little or no effects on child outcomes, is of key economic policy relevance. Previous studies have failed to document such thresholds for non-clinical outcomes. However, earlier studies either use cross-sectional variation and/or examine effects at *much* higher levels of exposure. The Swedish setting is particularly compelling in search for a threshold, since the blood lead levels were much lower than in most other countries, including the US, already before the phase-out of leaded gasoline. Using a semi–parametric estimator, we show that below an estimated blood-lead level between 3 to 5 μ g/dL (depending on the specification), the link between lead exposure and adult outcomes becomes much weaker. To put our estimates into perspective, reducing blood lead from 10 to 5 μ g/dL, which corresponds to CDC's recently changed definition of elevated blood lead levels, taking the non-linear relationship into account implies a 10 percent increase in the probability of high-school completion and a 3 percentage point decrease in property crime convictions.

The analysis of the 1980/81 reform shows that: (*i*) the timing of the changes in outcomes across birth cohorts corresponds well with the sharp reduction in lead consumption following the reform, but (*ii*) children in neighborhoods *below the exposure threshold* are not affected by the reforms, whereas (*iii*) the outcomes of children in neighborhoods *above the exposure threshold* improve significantly. These results provide support for the validity of the main results.

Animal studies suggest that lead interferes with development, yet we know little about the pathways to later human capital outcomes, or if different types of skills and behavior respond more strongly to a given level of exposure. The specific channels are of obvious importance for efficient remediating investments. We provide direct evidence on two key channels highlighted in previous work: cognitive and non-cognitive skills. Interestingly, in this low-exposure setting, we find no significant or consistent pattern with respect to *cognitive skills*. However, *non-cognitive skills* follow the same non-linear dose-response pattern as human capital and crime outcomes. When examining the effects on the non-cognitive ("Big-Five" personality trait) skill sub-

components, we find that lead primarily influences Impulsiveness, Emotional stability (Neuroticism) and Conscientiousness.

Our study contributes to a large literature examining the effects of childhood lead exposure on human capital accumulation. Much of this literature focuses on the effects on grades, and cognitive or behavioral test scores in (primary) school using crosssectional research designs. There is less evidence on effects on adult outcomes and very few previous studies report estimates that are interpretable as causal. However, a few recent studies address the identification problems. Reyes (2007) uses state-level paneldata and shows that the gasoline lead content reductions in the United States which occurred in the late 1970s and early 1980s account for as much as 56 percent of the decline in violent crime observed during the 1990s. Reves (2014) uses a similar research design for a sample of about 8,000 children in the National Longitudinal Survey of Youth (NLSY), finding a strong positive effect of childhood lead exposure on measures of delinquency for children followed up to age 17. Billings and Schnepel (2015) exploit the substantial measurement error in capillary blood lead tests to study a sample of 120 children that was treated in a lead remediation program in Charlotte, NC, and find that behavioral and educational deficits previously associated with high levels of early-life exposure can largely be reversed by intervention. Aizer et al. (2016) use an individual-level dataset of children with low lead exposure in Rhode Island and show that policies aimed at reducing lead hazards in homes increased reading test scores at age 8. Gazze (2016) exploits similar state mandates and finds that the share of kids attending special education decreases. A series of recent studies have exploited historical data from the late 19th and early 20th century to look at the importance of lead in water pipes for infant mortality, cognitive ability, and homicide (Clay, Troesken and Hains 2014; Troesken 2008; Ferrie, Rolf and Troesken 2012; Fiegenbaum and Muller 2016).

Our most important contribution is that we examine the effects of low levels of exposure on adult outcomes, while the earlier work examines short-run (elementary school) outcomes or effects at *much* higher levels of exposure.¹³ The thresholds that

¹³ Reyes (2007) report that average childhood blood lead levels in the adult US population will have decreased from 10 μ g/dL in 2002 to less than 3 μ g/dL in 2018. Billings and Schnepel (2015) examines of the intervention that occurs at 10 and 20 μ g/dL. The cohorts studied in Reyes (2007) were estimated to have a blood lead level between 10 and 20 μ g/dL. The average predicted childhood blood level in Reyes (2014) is about 7.6 μ g/dL. Gazze (2016) examines the impact on the share of children with special education following reductions in the share of children with B-

we document (~ 3 to 5 μ g/dL) are informative about the effects of early childhood lead exposure on adult outcomes at levels that are still common in the US and in many other countries today. In addition, previous work tends to examine the outcomes separately, while we provide comprehensive results for a broad set of adult outcomes using the same approach and population-wide administrative data.

Of key importance for policy makers (and parents) choosing between remediating investments following early lead exposure, we also present evidence shedding some light on the impact on the inputs through which low levels of lead exposure affects adult outcomes. We find that low levels of early lead exposure do primarily seem to influence Externalizing behavior, Neuroticism, and Conscientiousness, rather than Cognitive skills. This is an important finding since grades and achievement test scores is widely used as measures of leads effect on *cognition* (e.g. Aizer et al. 2016), but achievement tests have been shown to be substantially explained by personality traits (Borghans, Golsteyn, Heckman, Humphries et al. 2015, 2011). Moreover, Almlund et al. (2011) show that Neuroticism and Conscientiousness are particularly strongly related to Externalizing behavior and Heckman et al. (2013) find that Externalizing behavior, in turn, is the key explanatory factor behind the crime reducing effects of the Perry Preschool program. Since non-cognitive skills are more malleable to intervention than cognitive skills (Francesconi and Heckman 2016), our results suggest that policies that particularly target non-cognitive behavioral traits may be more successful in compensating for the negative effects associated with low levels of early exposure to lead.

Our study is also related to the large literature on the effects of air pollution on health, human capital, and mortality (e.g. Chay and Greenstone 2003; Currie and Neidell 2005; Lüchinger 2009; Jans et al., 2014). Most air pollution studies have not been able to assess the effects of environmental policies on adult outcomes. Isen, Rossin-Slater, and Walker (*forthcoming*) is a notable exception, who assess the effects of early childhood exposure to Total Suspended Particulate matter (TSP) by exploiting the 1967 Clean Air Act on earnings. We contribute, first of all, by examining the effects of exposure not only on earnings but also on academic outcomes, and cognitive- and non-cognitive skills. In addition, TSP is a mix of numerous components, including

 $Pb>10\mu g/dL$ and $20\mu g/dL$. Aizer et al. (2016) study effects of low exposure (<10 $\mu g/dL$) but their analysis is limited to 3rd grade reading and math test scores.

heavy metals such as lead. It is still uncertain, but highly policy relevant, what properties of particulate matter that are responsible for the toxic effects on health (Harrison and Yin 2000). We contribute by isolating the effects on adult outcomes of regulating a specific pollutant at low exposure levels that are still common on a global scale. This specificity should help policy makers find cost-effective solutions irrespective of the exact source of lead exposure.

The rest of this paper is structured as follows. Section 2 presents a conceptual framework on how to think about the consequences of lead exposure for child development. Section 3 discusses the measurement of lead exposure and describes our data. Section 4 presents the empirical results and Section 5 concludes the paper.

2 Conceptual framework

Lead is highly toxic to humans, irrespective of how it enters the body, and affects brain-development and organ functioning. Appendix B provides a review of the evidence on the specific mechanisms that affect development documented in animal studies. Below, we highlight some findings that are particularly important for our empirical strategy and the interpretation of our results and then use the model of human capital skill formation technology developed by Cuhna and Heckman (2007) to formalize how even low early lead exposure may affect adult outcomes.

First, early lead exposure poses the greatest risk because of a higher absorption rate into the blood among children and also because conditional on absorption, lead is more likely to affect the developing nervous system than the mature brain (Lidsky and Schneider 2003).

Second, blood-lead levels follow an inverted v-shaped pattern between the ages of 6 to 60 months, reaching its peak at the age of 24 months due to the intense hand-mouth activity at these ages (Needleman and Bellinger 2001; Canfield et al. 2003).

Third, early childhood lead exposure/lead take-up does not differ by gender,¹⁴ yet as we show empirically, lead exposure seems to affect boys more than girls. Several recent studies have similarly documented similar higher vulnerability of males to adverse prenatal, childhood, family, and neighborhood conditions.¹⁵ However, the direct

¹⁴ Studies that have investigated gender differences in blood lead levels have found that up until around the age of 10, the blood lead levels in boys and girls are highly similar (c.f. Strömberg et al. 1995).

¹⁵ Nilsson (2008); Chetty et al. (2016); Author et al. (2015); Grönqvist, Niknami and Robling (2016).

evidence on gender differences in the effects of early childhood lead exposure is mixed. Burns et al. (1999) find no gender differences in the impact on cognition, while Cecil et al. (2008) document that early childhood lead exposure is correlated with a lower brain volume using MRI scans at ages 19–24 years, particularly among boys, and particularly in the prefrontal cortex. Damage to the prefrontal cortex has been linked to the inability to plan and see the future consequences of one's actions, impulsive behavior, lack of self-control and a lower ability to delay gratification. Such non-cognitive traits have previously been linked to, for instance, a higher risk of convictions (Åkerlund et al. 2016). However, as for the more common measures of development, it is unclear whether the gender differences detected by the MRI scans are due to biological differences in the sensitivity to lead or simply reflect correlated unobservables. Moreover, both studies focus on children with high levels of exposure (>10 μ g/dL).

To better understand how early life lead exposure may influence individuals' life trajectories, it is useful to follow the framework proposed by Cunha and Heckman (2007, 2009) and characterize an individual's skill set as $\theta_t = (\theta_t^C, \theta_t^N)$, where θ_t^C is a vector of cognitive capabilities and θ_t^N is a vector of non-cognitive capabilities at age *t*. Adult outcomes is produced by a combination of the different skills at the end of childhood, with different weights placed on different outcomes. As suggested by Cunha and Heckman, the stock of skills in the next time period can be seen as a multi-stage developmental process where the stock of skills at age *t*+1 (θ_{t+1}) depends on the level of skills in the previous period *t*, investments (I_t) and parental environment (θ_t^P):

$$\theta_{t+1} = f_t \ (\theta_t \ , I_t \ , \theta_t^P).$$

The technology f_t is assumed to be increasing in each argument, twice differentiable, and concave in I_t . A key prediction of the model is that of "self-productivity", which arises when higher stocks of skills in one period create higher stocks of skills in the next period, i.e. $\partial f_t / \partial \theta_t > 0$. Secondly, the model assumes that the stocks of skills acquired by the end of the period t - 1 make investments in period t more productive, i.e. $\partial^2 f_t / \partial \theta_t \partial I_t > 0$.

Following Cunha and Heckman (2007), we assume that the technology can be depicted as a CES and that there are two childhood periods and one adult period in an

individual's life-cycle: early childhood (t=1) and late childhood/adolescence (t=2). An individual's adult outcome Y can then be written as

$$Y = f_2(\theta_1^C, \theta_1^N, \theta_P, \left[\gamma \tilde{I}_1^{\phi} + (1-\gamma) I_2^{\phi}\right]^{\frac{1}{\phi}})$$

where $\in [0,1]$, $\phi \in (-\infty,1]$. The share parameter (γ) is here a skill multiplier, and $\tilde{I}_1 = I_1 + \mu$, where μ is an exogenous shock occurring in period 1 (i.e. lead exposure). The elasticity of substitution, $1/(1 - \phi)$, is a measure of the substitutability of \tilde{I}_1 and $I_2 \cdot \phi$ represents the degree of complementarity/substitutability and determines how easy it is to compensate for low levels of investments (or a negative μ) in early childhood and in adolescence. When ϕ is small, it is difficult to compensate for low early investments (\tilde{I}_1) later in childhood (I_2). When $\phi = 1$, i.e. when \tilde{I}_1 and I_2 are perfect substitutes, the timing of investments is irrelevant. In the other extreme case, $\phi \rightarrow -\infty$, it is impossible to compensate for low early childhood investments later in childhood.

Early childhood is a sensitive period and the "neurological and behavioral effects of lead are believed to be irreversible" (WHO, 2016). Hence, $\phi < 0$ is likely to be the empirically relevant case. If so, even small early shocks may generate large negative effects on adult outcomes. Following Nilsson (forthcoming), to allow for differential effects of lead exposure across gender in (2) we allow ϕ to vary by gender, with $\phi_{boys} < \phi_{girls}$. Under this parameterization, the elasticity of substitution across periods is lower for boys than for girls, implying that it is more difficult to remedy an early shock for boys than for girls. Early lead exposure could therefore result in larger effects for males than for females.

3 Background

3.1 Measuring lead exposure

Lead started to be added to gasoline in Sweden in 1946 (Danielson, 1970). In response to the increasing trend in lead emissions from motor vehicles (see Figure 1), Sweden initiated a gradual phase-out of leaded gasoline in order to protect the environment and public health in 1970. While the 1970 reform prevented further increases in gasoline lead content, the main reductions in gasoline-lead levels occurred between 1973 and

1981 when the maximum amount dropped by 79% (see Figure 1). Gasoline lead was the main source of lead exposure in the general population and, as shown in Figure 1, there was a sharp decrease in children's blood-lead levels (B-Pb) from the 1970s until the mid-1990s when leaded gasoline was banned.¹⁶





Notes: From the left to the right, the figure shows the observed average lead levels in gasoline for the period 1963–1967, and 1969 (hollow squares) (c.f. Danielson 1970), the timing of the first reform (vertical line), the maximum allowed lead content in gasoline for premium grade (hollow circles) and regular grade gasoline (solid circles), before 1980 the regulation was the same for all grades (data from The SPI). The right-hand side y-axis shows the blood-lead level of elementary school children (μ g/dL) (c.f. Stromberg et al. (1995) and Stromberg et al. (2003).

To measure local lead exposure levels, we use data from the Swedish Environmental Protection Agency which since 1975 has monitored heavy metal air pollution every fifth year using a nationwide grid of moss (bryophytes) samples. The method of using mosses as biomonitors of heavy metal air pollution was developed in Sweden in the 1960s in pioneering work by Rühling and Tyler (1968, 1969) and is by now well

¹⁶ In Sweden, lead water pipes were never an issue and lead based paint was banned in the early 1920s. Even after the lead reforms in the 1980s, 80% of the air lead levels were still due to traffic (MOENR 1994).

established.¹⁷ Moss is particularly suitable for biomonitoring of air pollution levels for several reasons. The lack of roots implies that moss solely depends on surface absorption of pollution through precipitation and dry deposition of airborne particles. Mosses can be found in abundance in nearly all environments. The absorption and retention of metals is high, and it is possible to distinguish temporal patterns in pollution levels.¹⁸

Biomonitors have some advantages over regular pollution monitors: The main being its simplicity, accuracy, and low cost which allow a large number of sites to be included in the surveys. In the Swedish moss survey, samples from around 1,000 locations are collected. Additionally, unlike regular pollution monitors which often go in and out of operation as a response to changes in local pollution levels, the samples are collected all over Sweden using a systematic procedure.¹⁹ The growth segments over the last three years are pooled and analyzed and provide a measure of the local air lead levels during the three preceding years.

We use the samples collected in 1975, 1980 and 1985, which reflect the average lead deposition levels during the years 1972–1974, 1977–1979, and 1982–1984. The selection of these years is made for two reasons. First, between these years the maximum allowed grams of lead per liter of gasoline decreased sharply. As shown in Figure 1, the maximum allowed lead level in gasoline was reduced from 0.7 to 0.4 g/L (2.65 g/gal to 1.51g/gal) in 1973. This change applied to all types of gasoline. In 1980 (1981), there was an additional cut in allowed lead levels for regular (premium) gasoline from 0.4 to 0.15 g/L (0.56 g/gal). In 1986 (1995), lead was banned for regular (premium) gasoline. Second, the sample is restricted to cohorts who have reached adulthood and for whom the exposure in early childhood is observed.

The correct location of the specific sampling sites and the collection procedure are strictly defined. However, the sampling locations are not always identical across years. Hence, to construct a measure lead exposure for each period and neighborhood, we calculate the inverse distance weighted average air lead level for each parish using the

¹⁷ On a national scale, the use of moss as air pollution monitors expanded to Norway and Finland in 1985. Since 1995, 28 countries participate in a bi-decennial survey designed to study regional differences and time trends in heavy metal deposition throughout Europe.

¹⁸ The annual growth segments are easily distinguishable and the transportation of metal across segments is minimal.

¹⁹ The sampling sites are chosen with care; they should be located at least 300 meters away from bigger roads and closed residential areas, or at least 100 meters from smaller roads and single houses. At each sampling site, 5 to 10 subsamples are collected in an area of 100 m^2 .

population weighted parish centroids and the 10 closest sampling sites. The average parish in our data have about 3,000 inhabitants, i.e. it is slightly larger than US Census Block Groups.

As we show in Online Appendix A, there is a great deal of variation in the local lead levels between neighborhoods. Figure A1 and Figure A2 presents maps for the lead concentrations in the neighborhoods in 1975 and 1985, using our definition of exposure. From these, it is clear that the moss lead concentrations fell sharply between 1975 and 1985. Moreover, Figure A3. shows that the entire distribution of lead levels shifted dramatically between 1975 and 1985 and became more compressed. Figure A4. shows the distribution of within-neighborhood changes in lead exposure between 1975 and 1985. We can see that most neighborhoods experienced decreases in moss lead levels by between 15 and 45 mg/kg, with an average reduction of 30 mg/kg. A variance decomposition reveals that most of the variation in lead exposure is due to within neighborhood differences in the reduction of early childhood lead exposure across cohorts constitute a key feature of our research design.

Three important questions regarding the local lead exposure definition should be addressed before proceeding with the empirical analysis. The first question concerns the arbitrary choice of using the 10 nearest sampling sites to define the neighborhood of birth lead exposure. To test the sensitivity of the analysis to this assumption, we have also used the five nearest sample points instead. The differences between these definitions are small and they are highly correlated ($\rho = .99$). Second, to cross-validate our approach, we reestimate the level of lead at each sampling point, as opposed to neighborhood, after excluding the sampling point of interest. We do this for all sample sites, and then correlate the actual and the predicted air lead levels. The correlation is high ($\rho = .82$), indicating that the interpolation method provides reasonably accurate predictions of local air lead levels.

Finally, as in any study using data on local pollution levels rather than individual exposure, an important question is how well the lead levels in moss predicts the actual B-Pb levels in children. Unfortunately, there exist no data that monitors the trends in B-Pb levels among young children or the population in general in Sweden during this time

²⁰ The overall standard deviation for our measure of lead exposure is 17.93 while the between and within standard deviations are 10.78 and 14.35, respectively.

period. However, since 1978 venous blood samples have been collected from about 120 primary school children (age 8–10) per year in two municipalities in southern Sweden.²¹ One of these municipalities (Landskrona) has also collected around 50 moss samples using the same procedure as in the national survey at three time points (1984, 1995 and 2006). Most previous studies using aggregate data on pollution have been forced to assume that local air pollution exposure is a valid proxy for actual exposure. However, the two datasets from Landskrona provide a unique opportunity to assess the strength of the relationship between measures of local air lead exposure and measures of children's lead exposure. Figure 1 shows the mean B-Pb levels among primary school children in 1976–2000. Nilsson et al. (2009) link the average lead level of the nearest moss samples to the children using their home addresses. Controlling for important individual characteristics, time and neighborhood fixed effects, they estimate a Blood-Moss-lead elasticity for the pre-ban period (i.e. before 1995) of 0.44. This implies that a 10% reduction in Moss-Pb corresponds to a 4.4% decrease in primary school children's B-Pb, and suggests that the drop in air-lead exposure between 1982 and 1994 can account for as much as 50% of the change in children's B-Pb levels. Appendix B provides more details on the data and estimations.

However, it is important to remember, that the relationship between environmental lead exposure and young children's blood lead levels is significantly higher than among older children due higher exposure and higher absorption. For example, Reyes (2007) finds that the elasticity between lead in gasoline and B-Pb in children aged 0–6 is around 30% higher than among children aged 6–12. In this analysis, we therefore adjust the estimated elasticity accordingly when estimating the relationship between adult outcomes and early childhood B-Pb levels.

3.2 Micro data

Our data originate from three distinct databases at the Institute for Labor Market Policy Evaluation (IFAU), the Department of Economics at Uppsala University and the Institute for International Economic Studies (IIES) at Stockholm University. All databases draw from the same population based registers maintained by Statistics Sweden (SCB), which contain the information necessary to identify our main sample as well as a range of standard background characteristics such as demographics and parental socio-

²¹ The methods and results for the trends in childhood lead exposure are described in detail in Strömberg et al. (2003).

economic status. However, the databases differ in terms of the outcome measures included.

In the main analysis, we include all individuals born in Sweden in the three years prior to the year in which the moss samples were collected; that is all those born in 1972–1974, 1977–1979 and 1982–1984. The reason for the 1972 and 1984 constraint is that many of the individuals born after 1985 are less likely to have finished schooling at the time we measure their outcomes and that the first lead exposure measure available reflects the situation in 1972–1974 (i.e. mosses collected in 1975). As explained above, the lead levels in the mosses measure the local lead deposition at ages 1-3 years, which corresponds to a particularly sensitive period in human development and a period with high lead uptake rates. We limit our sample to children who have completed compulsory schooling (9 years of education) and who were born in Sweden, so that their neighborhood of birth (and hence, childhood lead exposure) is known. Hence, in our main analysis, we focus on outcomes for which we are able to measure at the same age for all cohorts.²² Human capital is measured using compulsory school grade point average (GPA) (age 16) and the probability of graduating from high school (age 19). We percentile rank the GPA to account for changes in the grading system over cohorts. We observe both these outcomes for all individuals until 2013.

We measure criminal behavior using administrative data on criminal convictions between 1985 and 2008 provided by the National Institute for Crime Prevention (BRÅ). As is typical with administrative crime data, we cannot directly observe criminality, but rather use convictions as a proxy for criminal activity. The data includes full records on the type of crime as well as the exact date of the offense for all individuals up to the age of 24.²³ A conviction may include several crimes and all crimes are recorded. Speeding tickets, and other minor crimes not severe enough to warrant a trial, are not covered. We study crime as the probability of being convicted at least once (for any crime) and the more specific and most common crimes categories: violent crime (representing the full spectrum of assaults from minor assault to murder) and property crime (representing the full spectrum of thefts from shop-lifting to robbery).

 ²² Sweden is one of the countries in the OECD with the highest age when obtaining the first university-level degree (average age 29) and more than 40 percent of each birth cohort also enrolls in college (OECD 2014).
 ²³ The age of criminal majority in Sweden is 15, meaning that we cannot observe criminal convictions before the age

 $^{^{23}}$ The age of criminal majority in Sweden is 15, meaning that we cannot observe criminal convictions before the age of 15.

We shed some light on the mediating factors affecting the adult outcomes by also studying the effects on cognitive and non-cognitive skills as measured for men at military enlistment. All men were obliged by law to go through the military draft. However, due to reforms in the military enrollment procedures affecting those born during the 1980s, the enlistment outcomes are only used for those born before 1980. Among these cohorts, about 90 % of all men went through the draft at the age of 18 or 19.²⁴

The cognitive test scores are Stanine (Standard Nine) test scores which are similar to the AFQT in the US. The score is an evaluation of cognitive ability based on several subtests of logical, verbal and spatial abilities and a test of the draftees' technical understanding. The results on these subtests are combined to produce a general cognitive ability ranking on a 1–9 scale. The test score is standardized within each cohort of draftees to account for any minor changes in the tests over time.²⁵

The non-cognitive score is based on a 20– to 25–minute interview with a psychologist resulting in four different scales, all ranging from 1 to 5, here displayed with their respective sub-categories and the Big-Five traits of Personality²⁶ (in italics) to which they are related: (i) Social maturity (extraversion (*E*), having friends (*E*), taking responsibility (*C*), independence (*O*)), (ii) Psychological energy (perseverance (*C*), ability to fulfill plans (*C*), to remain focused (*C*)), (iii) Intensity (the capacity to activate oneself without external pressure (*C*), the intensity and frequency of free-time activities (O)), and (iv) Emotional stability (ability to control and channel nervousness (*N*), tolerance of stress (*N*), and disposition to anxiety (*N*)). The non-cognitive sub-score tests are combined into a Stanine scale which we use as our main measure of non-cognitive ability.²⁷ Table A1. in the online appendix provides summary statistics for all outcomes.

4 Results

Empirically, we exploit the variation in lead exposure induced by the phase-out of leaded gasoline in the following three ways. Using the full data, we first provide (1)

²⁴ In principle, only the physically and mentally handicapped were exempt from the enlistment tests.

²⁵ The test has been subject to evaluation by psychologists and appears to be a good measure of general IQ (Carlstedt 2000).

²⁶ (O)peness to Experience, (C)onscientiousness, (E)xtraversion, (A)greeableness, (N)euroticism.

²⁷ Appendix D in Nilsson (2016) provides an account of how the military non-cognitive sub-scores (c.f. Bihagen et al., 2012) relate to the Big-Five traits of personality. Lindqvist and Vestman (2011) relate the cognitive and non-cognitive skills to labor market outcomes.

linear fixed effect estimates, and then provide results from (2) a semi-parametric fixed effects model that allows for a fully flexible exposure-response function. Finally, (3) we home in on the cohorts born between 1976–1984 and provide a detailed analysis of the effects of the large 1980/81 reforms through which the maximum lead content dropped from 0.4 g/l to 0.15 g/l.

4.1 Baseline results and robustness

To identify the effect of lead exposure on life trajectories, we use variations of the following linear regression model regressed on data aggregated to the cohort-neighborhood level:

(3)
$$Y_{nc} = \alpha + \beta Lead_{nc} + \theta X_{nc} + \delta Z_{mc} + \lambda_n + \lambda_c + \varepsilon_{nc}$$

where the Y_{nc} is compulsory school GPA (percentile rank), the probability of completing high school, or the probability of having been convicted for a crime up to the age of 24. *Lead_{nc}* is the local air lead levels measured in terms of milligram per kilogram (mg/kg) moss. This measure varies for individuals born in different years (*c*) and in different neighborhoods (*n*). X_{nc} and Z_{mc} represent vectors of parental and individual characteristics aggregated to the neighborhood level (*n*) (parental income and education, maternal age at birth, and gender, and municipality (*m*) level covariates (employment rate and population size). We include neighborhood fixed effects (λ_n) that account for persistent differences across neighborhoods over time, and birth cohort fixed effects (λ_c) which take cohort-specific effects into account. ε_{nc} is the error term. Throughout, we cluster the standard errors at the commuting zone (CZ) level to account for arbitrary serial correlation and heteroscedasticity. Clustering at the CZ level also accounts for spatial correlation in the errors between neighborhoods within the 74 CZs.²⁸

The parameter estimate of interest, $\hat{\beta}^{OLS}$, should be interpreted as the effect of an additional unit of lead exposure (1 mg/kg) in early childhood on adult outcomes. Before proceeding to the main results, we discuss three potentially important threats to consistent estimates of β in equation (3).

²⁸ There are 2,586 parishes in the data. Statistics Sweden divided these into 74 commuting zones. For a map of the 74 commuting zones, see: *http://www.scb.se/Statistik/AM/AM0207/_dokument/LA2009_KARTA.pdf*.

The first concern is that parents might sort into residential location based on factors associated with lead pollution levels. This could overstate the effects of lead exposure if parental preferences for clean air are also associated with higher parental investments in children. On the other hand, pollution tends to be higher in densely populated areas and, at the same time, metropolitan areas often attract highly educated parents with more resources, contain better access to quality child care, schools, health care and other amenities that are positively associated with the adult outcomes of the children. Such local amenities could, in turn, for instance, via gentrification, result in an underestimated role of childhood lead exposure if not properly taken into account.

When Tiebout sorting occurs, the most important predictor of both residential location and child outcomes is probably socioeconomic background. To better understand the importance, direction, and severity of this bias, we start by regressing parental educational attainments and income on their children's lead exposure.²⁹ This gives a hint of whether omitted variables associated with parents' characteristics, or migration that changes the composition of families within neighborhoods, are correlated with changes in lead exposure.

The correlation between lead exposure and parental characteristics are displayed in Table A2. Panel A shows the cross-sectional correlation and Panel B shows the same results after adding neighborhood fixed effects to the specification. Overall, the coefficients in Panel A suggest that higher lead exposure of the child is associated with lower parental education, lower pre-birth maternal income, and lower maternal age at birth. This suggests that cross-section regressions likely overstate the effects of lead exposure on children's outcomes. After taking into account fixed differences across neighborhoods, the coefficients generally drop substantially and often change sign.

²⁹ These parental characteristics can be considered to be predetermined in the present context since the lion's share of the parents (>95 percent) were born before 1960, i.e. before environmental lead exposure became a serious environmental problem in Sweden. Using mosses collected from 1860 until 1968, Rühling and Taylor (1968) show that in the southern part of Sweden (the most highly exposed in the present sample), the increase in lead concentrations in moss was restricted to two distinct periods: a first increase towards the end of the nineteenth century, and a second increase during the 1960s (80-90 mg/kg in 1968). Before this, the average lead level in Skåne (the southernmost region in Sweden with the highest lead level in the data used in this study) was around 40μ g/kg moss. They conclude that the first rise is probably due to industrial pollution, possibly due to the increased use of coal, and that the second rise is more than likely caused by the rapidly increased use of lead gasoline. Exactly the same pattern is found in a study by Rehnberg et al. (2000). They use extraordinary data on lead levels in lake sediments to examine regional trends in lead depositions in Sweden over a period of 4,000 years. In particular, the lead concentrations in the lake sediments increased by 50% between 1960 and the peak year of 1970. Hence, the parents of the children were exposed to relatively low levels during their own childhood and therefore, the lead levels during the parents' childhood are not expected to be able to influence the children's adult outcomes to any large extent.

However, note that neither the coefficients in Panel A nor in Panel B are statistically distinguishable from zero, and that parents' socioeconomic characteristics show no consistent association with their children's lead exposure.

A second concern is that although there is a consensus that the overall drop in lead exposure is driven by gasoline lead policies (see Thomas et al. 1999), local lead exposure could also proxy for unobserved neighborhood characteristics that are not captured by the neighborhood fixed effects. Perhaps the two most obvious candidates include changes in the local business cycle and population density, which can both be hypothesized to be linked via emissions to local lead levels and to outcomes. To the extent that they are correlated with changes in parental income and education, the analysis in Table A2. suggests that they are not a concern.

However, from the bi-decennial Censuses, we also gathered municipality level data on population density and employment covering not only the parents. Figure A5. and Figure A6. show that changes in local lead exposure between 1975 and 1980 and between 1980 and 1985 appear to be unrelated to changes in population density or local employment. Put differently, in terms of labor market conditions and migration patterns, places with high lead declines seem to have been relatively similar to places with low lead declines. Accordingly, as we see below, controlling for these variables in the regressions does not change the magnitude of the estimates in any meaningful way.

Similarly, local lead levels may be correlated with other pollutants that also affect child outcomes. The focus on the changes in air lead levels induced by government regulations that specifically targeted gasoline lead content should mitigate much of this concern. Still, since the moss data also hold information on other common heavy metals, it is easy to make an initial assessment of the potential severity of this problem. Of all the observable heavy metal pollutants in the data, only Cadmium (Cd) displays nearly as large and widespread changes during the observation period as does lead. Cd has previously been found to be associated with adverse health outcomes (kidney damage, bone disease).³⁰ The data also include information on levels of Copper (Cu)

³⁰ Early Cd exposure has been shown to be able to produce neurotoxic effects in laboratory experiments (Anderson et al. 1997; Peterson et al. 2004), and in utero exposure to cadmium may affect infant health in humans (Currie and Schmeider 2009). Air Cd and air Pb concentrations display a fairly high correlation at the parish level. Hence, changes in air Cd levels could at least partly explain the estimated relationship between lead and subsequent outcomes. However, in this context, it is not likely that the estimates for Pb are driven by the changes in local air Cd exposure rather than local air lead exposure. This is because unlike lead, the primary exposure route of cadmium is dietary rather than respiratory (WHO 1972; IPCS 1992; Moon et al. 2003; Ohlsson et al. 2005) Cd accumulates in

and Zinc (Zn). Cu and Zn have been suggested to be used as marker elements for motor vehicle emissions in previous studies and are therefore useful to include in the analysis to control for other traffic related pollutants.³¹ As we will see in the next section, the estimates are stable when controlling for these other heavy metal pollutants.

4.1.1 Baseline results

Table 1 presents the main results. Each table entry corresponds to a separate regression. The table provides results for GPA, High school completion, and Criminal convictions by type of crime. In the first row, we only control for cohort and neighborhood fixed effects. Then, we successively add more controls. Our preferred specification is found in the bottom row and provides the neighborhood fixed effect estimates of increasing moss lead by 10 mg/kg, after including the full set of controls (individual, parental, municipal and other traffic-related pollutants).

The pattern of the coefficients is clear. Higher levels of exposure in early childhood lead to lower GPA and high-school completion, and a higher risk of being convicted. Overall, adding the controls hardly changes the estimated coefficients at all. The exception is the effect on violent crime, which is always positive but becomes insignificant in the full specification. The property crime coefficient is positive and stable across the different specifications but is always statistically insignificant in this linear specification. We will return in Section 5 to the question of how to assess the magnitude of these estimates.

crops, fish and livestock. But since only a small proportion of the food that children in Sweden (and elsewhere in most developed countries) eat is locally produced, it is *a priori* not expected that the intertemporal changes in *local* air Cd levels in early childhood are necessarily associated with adverse future outcomes.

³¹ Brake linings wear are the major source of road traffic emitted Cu, and Zn concentrations are high in tiers (e.g. Johansson et al. 2009; Hjortenkrans et al. 2007).

Outcomes:	Grade 9 GPA (pct. rank)	P(High school complete)	P(Ever convicted)	P(Property crime)	P(Violent crime)
	(1)	(2)	(3)	(4)	(5)
Baseline	1890**	0034**	.0014**	.0007	.0004**
	(.0961)	(.0014)	(.0007)	(.0008)	(.0002)
+ individual & family	2018***	0038**	.0017**	.0008	.0005**
controls	(.0716)	(.0015)	(.0006)	(.0007)	(.0002)
+ municipality	2009**	0039**	.0018**	.0008	.0005**
characteristics	(.0716)	(.0015)	(.0007)	(.0007)	(.0002)
+ other traffic	2097**	0031**	.0018**	.0009	.0002
pollutants	(.0925)	(.0014)	(.0008)	(.0009)	(.0003)
Mean of Outcome	[50.72]	[.79]	[.164]	[.070]	[.025]
Neighborhood FEs	Yes	Yes	Yes	Yes	Yes
Year of birth FEs	Yes	Yes	Yes	Yes	Yes

Table 1. The effect of childhood lead exposure on life trajectories

Notes: The table presents the coefficient on early lead exposure from separate OLS regressions. Coefficients and standard errors are scaled by a factor of 10. Crime (in columns 3-5) is defined as having been convicted at least once up to the age of 24. The sample consists of children born in 1972-1974, 1977-1979 and 1982-1984. Childhood lead exposure is measured as an average over ages 1-3. The baseline model only controls for cohort and neighborhood fixed effects. We then add individual (gender, birth month dummies, birth weight) and parental controls (dummies for the unique combination of the parent's highest completed level of education (55 combinations), maternal age dummies in years at birth, dummies for quartile of total parental earnings two years before birth (in 1985 for crime outcomes), birth order dummies. Municipality controls include population size and employment rate from the 1975, 80 and 85 censuses. Other traffic related pollutants include Cadmium, Zinc and Copper. Sample means expressed in percent are shown in brackets. Cluster robust standard errors (at the commuting zone level, 74 cells) are shown in parenthesis. *** = significant at 1 %, * ** = significant at 5 %, * = significant at 10 %. The preferred (full) specification is highlighted in bold.

4.1.2 Estimates by Gender

As discussed in Section 2, several recent studies have documented a greater vulnerability of males to adverse early conditions, but the evidence on gender differences in the impact of lead is mixed. Hence, it is of interest to examine differences across genders in our low lead setting. Panel A of Table 2 repeats the results from the full sample from the main specification in Table 1, while panels B and C provide the same estimates separately for boys and girls. It is striking that the baseline results not only for criminal convictions but also for GPA and high school completion seem to be driven by the impact on boys. Since early childhood lead take-up does not differ across genders, these results suggest that the effects of early lead exposure on adult outcomes are stronger for males than for females.

Outcomes:	Grade 9 GPA (pct. rank)	P(High school completion)	P(Ever convicted)	P(Property crime)	P(Violent crime)
	(1)	(2)	(3)	(4)	(5)
	Panel A: Full sar	nple			
Early lead exposure	2097** (.0925)	0031** (.0014)	.0018** (.0008)	.0009 (.0009)	.0002 (.0003)
Sample mean	50.72 Panel B: Girls	.79	.164	.070	.025
Early lead exposure	1452 (.1110)	0011 (.0015)	.0004 (.0007)	.0001 (.0006)	.0000 (.0001)
Sample mean	56.69 Panel C: Boys	.808	.079	.045	.006
Early lead exposure	2773*** (.0989)	0051*** (.0019)	.0029** (.0013)	.0017 (.0014)	.0006 (.0005)
Sample mean	44.85 Panel D: Low Inc	.769 come Parents	.244	.095	.042
Early lead exposure	2640** (.1170)	0037* (.0019)	.0017 (.0011)	.0014 (.0011)	.0007** (.0004)
Sample mean	45.39 Panel E: High Ind	.739 come Parents	.191	.087	.033
Early lead exposure	2120** (.1020)	0027** (.0013)	.0015* (.0007)	.0004 (.0007)	0002 (.0002)
Sample mean	55.38	.834	.136	.054	.016

Table 2. Heterogeneity	in the effect of lead	exposure by o	gender and	parental income

Notes: The table presents the coefficient on early lead exposure from separate OLS regressions. Coefficients and standard errors are scaled by a factor of 10. Crime (in columns 3-5) is defined as having been convicted at least once up to the age of 24. The sample consists of children born in 1972–1974, 1977–1979 and 1982–1984. Childhood lead exposure is measured as an average over ages 1–3. Parental income is measured as the sum of each parent's earnings. Low(High) income parents are classified as below(above) median family earnings either two years prior to when lead exposure is measured (for human capital outcomes) or in 1985 (for crime outcomes). Besides neighborhood and birth cohort fixed effects, the model controls for the full set of controls as specified in the note of Table 1. Cluster robust standard errors (at the commuting zone level (74 cells)) are shown in parenthesis. *** = significant at 1 %, * ** = significant at 5 %, * = significant at 10 % .

4.2 The role of nonlinearities

The default dose-response model in toxicology is the threshold model, which posits that the effect of toxins follows a hockey-stick shape across the distribution. Below certain thresholds, further reductions in exposure no longer affect the outcome of interest. Some indicative health thresholds for lead have been established for children: encephalopathy, seizures and coma (>70–80 μ g/dL), renal (kidney) failure (50 μ g/dL), hemoglobin disruption (25 μ g/dL) (Agency for Toxic Substances and Disease Registry 2007). However, no thresholds have been established for more subtle effects on neuro-cognitive development or behavior. To what extent that this is due to the limitations of the cross-sectional research designs, or that previous studies generally test for non-linarites at high levels of exposure, or both is unclear.³² The literature that attempts to address the identification issues and directly test for non-linearities has not detected any thresholds, but again, it has examined effects at much higher exposure.³³

To examine the role of non-linearities in our low-exposure setting, we estimate semi-parametric fixed effect models (Baltagi and Li 2002). This estimator takes into account neighborhood fixed effects and the other controls used in the linear specification, but allow the exposure-response function to be estimated fully flexibly:

(4)
$$Y_{nc} = \alpha + f(Lead_{nc}) + \theta X_{nc} + \delta Z_{mc} + \lambda_n + \lambda_c + \varepsilon_{nc}.$$

For human capital outcomes, the estimated exposure-response function, $\hat{f}(\cdot)$, for males is presented in Figure 2 together with bootstrapped 95% confidence intervals (1,000 replications).³⁴ The top panel shows the results for GPA which reveal some indication of non-linearites at the very bottom of the distribution (~ 20 mg/kg), where the relationship flattens out. However, the confidence bands are wide, and it is impossible to reject that the relationship is linear for GPA. The pattern is much clearer for high-school graduation. Here there is a clear threshold at around 30 mg/kg below which further reductions in exposure have limited or no effects on graduation rates.

³² There is no strong theory that could give any guidance on the exact location of the thresholds for the outcomes we consider (c.f. Needleman 2004). In fact, some cross-sectional studies find that the effects of lead are *increasing* at lower levels of exposure, i.e. changes in exposure at low levels of exposure have stronger effects on test scores than changes at high levels of exposure (see Canfield et al. 2003; Skerfving et al. 2015). The unexpected direction of the "nonlinear dose-response curve is a mystery" (Skerfving et al. 2015, p.118). A potential explanation is that at lower levels of exposure, the composition of children that are exposed to lead is likely to differ from the composition of children at higher levels of exposure. To give a concrete example, before the ban of leaded gasoline, more or less all children were exposed to some extent, with gasoline lead being the key source of exposure. After the phase-out, only a share of the children are exposed, namely children living in low standard housing with deteriorating water pipes, lead paint walls, or living close to heavy industries like smelters. Therefore, it is possible that the bias from unobserved factors is stronger in a low lead setting than in high lead setting, generating a steeper dose-response curve in settings where the sources of exposure are more family specific and hence likely to more closely reflect (unobserved) family characteristics.

³³ Reyes (2011) directly examines non-linarites in the effects of early lead on violent crime, but does not find any threshold effects at average blood lead levels between $10 \,\mu g/dl$ and $20 \,\mu g/dl$.

 $^{^{34}}$ To ease the computational burden, we collapse the individual level data at the neighborhood-by-measurement year level and weight the regressions by the number of individuals in each cell. The non-parametric part is estimated using a narrow bandwidth (.3) robust local linear estimator (LOWESS). To further reduce the influence of outliers, we follow standard practice and we exclude the bottom/top 1% of the observations in the exposure distribution.

Above 30 mg/kg, there is a clear and statistically precise negative relationship between exposure and the probability of graduating from high school. For completeness, Figure A7. provides the same estimates for females. For females GPA, there is a tendency to a non-linear effect at the very top of the exposure distribution, but overall, as expected from the linear regression results, the pattern is less clear for females than for males.

Figure 3 provides results for Criminal convictions. For Any crime, there is a tendency to the same pattern as for high school completion, but the relationship is imprecisely estimated. For Property crime, the relationship is clearer, above approximately 65mg/kg there is a clear and statistically significant positive relationship between lead exposure and property crime. Below 65mg/kg, the regression line is flat: changes in lead exposure have no impact on Property crime below the threshold. For Violent crime, there is a tendency to a similar pattern as for Property crime, but it is not statistically significant.

The semi-parametric regressions suggest that 65 mg/kg could be a candidate for a threshold for (property) crime. For human capital outcomes, the relevant threshold seems to be located at a lower level (20–30mg/kg). To verify the locations of these thresholds, we follow standard practice and estimate a series of piece-wise linear equations that allows for a change of slope at different *assumed* thresholds (Hansen 1999). Based on these regressions, Hansen's Likelihood Ratio (LR) test statistic provides "no-rejection regions" containing the best estimate of where the *true* threshold is located. Appendix A Figure A8. and Figure A9 show the LR-statistic threshold locations. In general, they verify the location suggested by the more flexible semi-parametric regressions.³⁵

³⁵ Following Hansen (1999), the criterion to locate the true threshold is the assumed threshold at which the sum of squared residuals in the piecewise linear regression is minimized. Hansen's likelihood ratio (LR) statistic is $nT \frac{SSR_s - SSR_{min}}{SSR_{min}}$ where SSR_s is the sum of squared residuals at the assumed threshold *S*, SSR_{min} is the minimum value of this statistic across all thresholds, *n* is the number of observations and *T* is the number of time periods. The advantage of the LR statistic over the *SSR* is that its asymptotic distribution (critical-value) is known.



Figure 2. Semi-parametric estimates for human capital

Notes: The graphs present estimates along with a 95 % confidence interval (bootstrapped with 1000 repetitions) from semiparametric fixed-effects models (see Baltagi and Li 2002). The dependent variable in the top graph is grade 9 GPA and in the bottom graph P(High school completion). The sample consists of males born in 1972-1974, 1977–1979 and 1982–1984. Childhood lead exposure is measured as an average over ages 1–3. See note Table 1 for the full specification set of controls.



Figure 3. Semi parametric estimates for crime

Notes: Semi-parametric fixed effects estimates (see Baltagi and Li 2002) with 95% bootstrapped confidence intervals (1,000 repetitions). The dependent variable in the top graph is P(Ever convicted), in the middle graph P(Property crime) and in the bottom graph P(Violent crime). Additional details on the data are given in the notes to Figure 2.

In summary, the analysis in this section reveals clear signs of nonlinearities in the effect of early childhood lead exposure on adult outcomes. Since the average blood-lead level in the cohorts considered were already initially lower than 10 μ g/dL, these results also suggest that, if anything, the threshold of the adverse effects of lead exposure on adult outcomes lies at a level that is significantly lower than a B-Pb level of 10 μ g/dL.³⁶ In Section 5, we use the elasticity provided by Nilsson et al. (2009) to estimate the childhood B-Pb level corresponding to the moss lead level documented in this section.

4.3 Isolating the effects of the 1980/81 reform

It is well documented (c.f. Thomas et al. 1999) that the general dramatic drop in lead exposure that occurred during the observation period is due to the policies mandating reductions of the maximum lead contents of gasoline. However, one may still be concerned that at least a part of the local variation in lead that is used in the panel analyses is driven by correlated unobservables that are not captured by either the neighborhood-of-birth fixed effects or the extensive set of control variables.

To further try to address this concern, below we provide results from an analysis that exploits the timing of the 1980/81 reform.³⁷ As shown in Figure 1 above, the maximum lead content of regular (premium) gasoline decreased from 0.4 to 0.15 grams per liter gasoline in January 1980(81).³⁸ Combining this information with yearly municipality data on gasoline sales by grade (available for the period 1976 to 1984), Figure A10. shows the average of the maximum gasoline lead sales (in metric tons) per municipality and year around the time of the reform. The sharp reductions in maximum gasoline lead in 1980/81 (a 63 percent drop) seen in Figure A10. provides an opportunity to more clearly isolate the effect of the reform and how the reform effect evolves across birth cohorts. We estimate the following model for children born between 1976 and 1984:

³⁶ See Appendix B for evidence on the pre-phase out B-Pb levels.

³⁷ We focus on this reform since first since there are several years since the previous reform was implemented (1973), and there are several years until the next reform was implemented (1986). This gives us a reasonably long pre-reform period, which we can use to examine pre-reform trends in the outcomes. It also enables us to examine the same outcomes that we focus on in the main analysis. Second, the 1980/81 reforms allows us to use a well-defined pre-reform measure of exposure (1977-1979). The 1973 reform is not as useful since the closest moss lead exposure measure captures a mix of pre and post-reform levels of exposure (1972-1974). The 1986 reform is also less useful both since it targeted only regular grade gasoline which implies that the changes in lead exposure are much more gradual than following the 1980/81 reforms which targeted all grades.

³⁸ As shown in Figure 1 children's blood lead levels dropped by 30% between 1979 and 1983. In a repeated sample of adults, the blood-lead levels in Stockholm dropped by 30% between 1980 and 1983 and air lead levels between 1979 and 1982 dropped by 50% in Stockholm (Elinder et al. 1986).

(5)
$$Y_{nc} = \alpha + \sum_{c=1976}^{1984} \gamma_c (\lambda_c \times \text{Lead }_n^{\text{Pre-Reform}}) + \theta X_{nc} + \delta Z_{mc} + \lambda_n + \lambda_c + \varepsilon_{nc}$$

where $Lead_n^{Pre-reform}$ is the local air lead levels just before the reforms (1977–1979), which we interact with the birth cohort dummies (λ_c). The parameters of interest are the cohort-specific coefficients (γ_c). In essence, these coefficients provide estimates of the effect of the reform across areas with different predicted changes in lead exposure generated by the reform. Using the predicted reform-induced variation in exposure, rather than the actual exposure, removes the potential confounding by unobserved factors that may determine actual exposure and affect children's adult outcomes. In this analysis, we are also able to directly control for local gasoline sales. We use the children born in 1978 as the reference cohort since they are aged above 3 at the time of the full reform and have hence passed the critical age (24 months) when lead take up and B-Pb levels start to decrease (Canfield et al. 2003).

However, we do take our earlier findings into account in this analysis. In particular, before the 1980/81 reforms, the median pre-reform exposure level is 29 mg/kg (see Figure A3.). Hence, at least for the human capital outcomes, the estimated threshold documented in Section 4.2 is well within the boundaries of the pre-reform exposure levels. Accordingly, we provide separate estimates of equation (5) for samples above and below the estimated human capital threshold (30 mg/kg). This provides us with four predictions on how the γ_c coefficients should evolve across the birth cohort, depending on the outcome and the pre-reform lead exposure in the neighborhoods of birth. We expect the human capital outcomes for:

- children in neighborhoods with pre-reform levels above 30 mg/kg to improve after the reforms, and
- children in neighborhoods with pre-reform levels below 30 mg/kg to remain unchanged after the reforms.

For Property and Violent crime the estimated threshold (65 mg/kg) is virtually out of range at the time of the 1980/81 reforms and hence, we expect that Property and Violent crime for

iii) and iv) children in neighborhoods above or below 30 mg/kg both should remain unchanged after the reforms.

We present the point estimate and point-wise 95%-confidence intervals for the γ_c 's (in Equation 5), together with joint *F*-tests of equality of the γ_c coefficients before and after the reform (i.e. standard difference-in-differences) for each outcome in Figure A11 and Figure A12. For brevity, we summarize these results by first standardizing the outcomes and plotting the coefficients from the four separate regressions for human capital outcomes (top) and crime outcomes (bottom) in Figure 4. The black dots represent point estimates of the γ 's for children in the above-exposure-threshold neighborhoods. The sold line is the local average of the coefficients (weighed by the inverse of the standard errors) in the above-threshold neighborhoods and the dashed line is the corresponding averages in the below-threshold neighborhoods.

The pattern for the different outcomes and exposure groups summarized in Figure 4 follows the outlined predictions. The trends in the coefficients for the cohort exposed to the reform at older ages do not differ before the reform between children in above and below threshold neighborhoods and are not statistically distinguishable from zero. For human capital, the coefficients in above and below threshold neighborhoods start to diverge for the 1979 birth cohort (age 1 at the first reform). *F*-tests of the pre vs. post reform coefficients (i.e. 1976–78 vs 1979–84) are statistically distinguishable at the 5%-level in the above threshold neighborhoods, but not statistically distinguishable in the below threshold neighborhoods. For Crime, as expected, the differences before vs. after the reform in the above and below threshold neighborhoods are small and not statistically distinguishable from zero with few exceptions.^{39 40}

 $^{^{39}}$ Two exceptions are worth mentioning. The first is the estimates for Property Crime for which the *F*-test suggests that there could be a reduction in crime in the low exposure areas, which is contrary to our expectations. For completeness, we also report the cohort-specific estimates for Any crime in Figure A13 despite the uncertainty of the location (if any) of the threshold and the failure to detect non-linearities (c.f. Figure 3 and the Hansen LR statistic for Any Crime in Figure A9). From Figure A13, we can see that after the reform, there is some indication that the probability of ever being convicted decreases following the reform in high-exposure areas but not in the low-exposure areas. This might suggest that for the more noisy Any Crime category, there exists a threshold at similar locations as for human capital outcomes. However, we interpret this finding for Any Crime with some caution given the impreciseness of the estimates in Figure 3 and Figure A9.

⁴⁰ For women, the coefficients mirror the main results. There are no clear effects of the 1980/81 reforms on women (see Appendix A Figure A14).

Figure 4. The estimated impact of the 1980/81 reform on human capital and crime



Notes: The figures summarize the result from estimates of equation (5). Here the outcomes are standardized before estimation (mean 0, SD 1) which allows for comparison across outcomes (See in Appendix Figure A11 and Figure A12 for full results). The black dots represent point estimates of Pre-reform lead exposure interacted with the cohort for children in high exposure neighborhoods and the hollow circles the estimates for children in low exposure neighborhoods (see text for details). The sold line is the local average (weighed by the inverse of the standard errors) of the coefficients from high exposure areas and the dashed line is the corresponding averages for children in low exposure areas using a bandwidth of 1 year. The dependent variables are for human capital (upper graph), compulsory school GPA and high-school completion and for crime (lower graph): P(Violent crime) and P(Property crime). The sample consists of males born in 1976-1984. Pre-reform lead exposure is the measurement reflecting exposure in 1977–1979. On top of the full specification controls (see note in Table 1) in this specification we are also able to control for municipality level gasoline sales (1000m³).

In summary, the timing of the changes in outcomes documented in the analysis in this section generally reinforces our beliefs that the main results on adult outcomes reflect changes in childhood lead exposure caused by the reforms. Importantly, if the results in our main analysis were due to unobserved differential trends across high and low exposure neighborhoods, we would not expect to find such a systematic pattern around the timing of the 1980/81 reform. The results provide support for the existence of threshold effects and for the differential impact of low levels of lead exposure across genders.

4.4 The mediating role of cognitive and non-cognitive skills

In this section, we directly examine the relevance of the two mediating channels highlighted in the conceptual framework: cognitive and non-cognitive skills. Distinguishing the role of these two sets of skills is interesting since it could provide additional supporting evidence for the internal validity of the main results, and serve as a comparison to effects of other early interventions (e.g. Heckman et al. 2013; Nilsson forthcoming). But it is also interesting for guiding the design of effective remediating policies following lead exposure: cognitive and non-cognitive skills differ in their malleability to intervention across the child's life-cycle. While cognitive ranks seem to stabilize around the age of 10, non-cognitive skills are malleable at least up to the early 20s (Francesconi and Heckman 2016). Hence, information on whether the effects on adult outcomes are mainly driven by cognitive or non-cognitive skills can provide important information about the appropriate types of intervention and the window of opportunity for remediating investments following early lead exposure.

Figure 5 provides the semi-parametric estimates for cognitive and non-cognitive skills. The top figure shows that there is no clear and consistent exposure-response pattern for cognitive skills. On the contrary, the bottom figure shows that for non-cognitive skills, the exposure-response pattern is similar to the non-linear shape documented for the adult outcomes. Below 30 mg/kg, the slope is not statistically distinguishable from zero, while above this level, it is consistently and significantly downward sloping.

It is also possible to go into even further detail by directly examining the effects on the four sub-scores that make up the overall composite non-cognitive score. Semiparametric estimates for each of the sub-scores are presented in Figure 6. The estimates in this figure suggest that the overall effects are driven by effects on psychological energy (Conscientiousness) and emotional stability (Neuroticism), and social maturity (Extraversion). These personality traits have previously been closely linked to externalizing behavior (e.g. aggression, hyperactivity, antisocial behavior) (Almund et al. 2011) and labor market outcomes, which is the key explanatory factor in the crime reducing effects of the Perry Pre-school program (Heckman et al. 2013).

4.5 Effects on labor market outcomes

Our analysis focuses on the outcomes for which the children in our sample are wellmeasured at the ages when we are able to observe outcomes for all cohorts. Within OECD, Sweden is one of the countries with the highest age when obtaining the first university-level degree (average age 29) and 40% of each cohort graduate from higher education (OECD 2014). This implies that measuring and examining the effects on labor market outcomes before the mid-30s is problematic, since many are still enrolled in higher education and hence only have a loose connection to the labor market.

With this caveat in mind, we examine the effects on earnings in Appendix D. In 2013, at ages 29–40, we find little evidence that early lead exposure, at the low-levels of exposure to lead that we observe, has any systematic effect on earnings. Future work could use the same analysis to examine effects on earnings at ages when current earnings better reflect permanent income. Böhlmark and Lindquist (2006) show that to obtain credible measures of permanent earnings require current income to be measured in the early to mid-30s for men and even later for women. To get a sense of the economic magnitudes, in the next section we do instead impute the predicted effects on mid-30s earnings using the estimated impact on high school completion.



Figure 5. Semi parametric estimates for cognitive and non-cognitive skills

Notes: The graphs present estimates along with a 95 % confidence interval (bootstrapped using 1000 repetitions) from semi-parametric fixed effects models (see Baltagi and Li 2002). The dependent variable in the top graph is cognitive skills and in the bottom graph non-cognitive skills. Both outcomes are standardized (mean 0, SD 1). The sample consists of males born in 1972–1974, 1977–1979. Childhood lead exposure is measured as an average over ages 1–3. Besides neighborhood and birth cohort fixed effects, the model controls for the full set of controls as specified in the note of Table 1.


Figure 6. Semi parametric estimates for components of non-cognitive skills

Notes: The graphs present estimates (males only) along with a 95 % confidence interval (bootstrapped using 1,000 repetitions) from semi-parametric fixed effects models (see Baltagi and Li 2002). The non-cognitive score is based on a 20- to 25-minute interview with a psychologist resulting in four different scales, all ranging from 1 to 5, here displayed with their respective sub-categories and the Big-Five traits of Personality ((*O*)peness to Experience, (*C*)onscientiousness, (*E*)xtraversion, (*A*)greeableness, (*N*)euroticism) to which they are related: (i) Social maturity (extraversion (*E*), having friends (*E*), taking responsibility (*C*), independence (*O*)), (ii) Psychological energy (perseverance (*C*), ability to fulfill plans (*C*), to remain focused (*C*)), (iii) Intensity (the capacity to activate oneself without external pressure (*C*), the intensity and frequency of free-time activities (O)), and (iv) Emotional stability (ability to control and channel nervousness (*N*), tolerance of stress (*N*), and disposition to anxiety (*N*) (see Nilsson 2016)). All outcomes are standardized (mean 0, SD 1). The sample consists of males born in 1972-1974, 1977-1979. Childhood lead exposure is measured as an average over ages 1-3. Besides neighborhood and birth cohort fixed effects, the model controls for the full set of controls as specified in the note of Table 2.

5 Discussion and economic significance

The existence and location of a moss-lead threshold value are interesting and policy relevant in themselves since at least 28 European countries measure local lead levels in moss with regularly, and many regions an cities also do so with more irregular intervals.⁴¹ But in order to attain a rough estimate of to which blood lead level the estimated moss -lead threshold value correspond, we use the model estimated in Nilsson et al. (2009) that relates moss lead to children's blood lead for the period preceding the ban of

⁴¹ The city of Portland is a recent example. *http://projects.oregonlive.com/air-pollution/heavy-metals/moss/ http://www.oregonlive.com/environment/index.ssf/2016/06/more_moss_studies_aim_to_uncov.html*

leaded gasoline (see Appendix B for a review of the method). Evaluating this model at the means of the independent variables, after adjusting the blood-lead moss-lead elasticity using the age-specific blood-lead gasoline-lead elasticity estimated in Reyes (2007), and under the additional assumption that the additive separable specification used in the estimation holds for both populations, the relevant blood lead level for property and violent crime (~65mg/kg) corresponds to an early childhood blood-lead level of about 5.7 μ g/dL. The corresponding figure for the high school completion threshold is approx. 3.5 μ g/dL.

However, it is important to keep in mind that we use the neighborhood lead levels as the exposure measure; hence, the documented threshold effect for crime associated with a *neighborhood* average blood-lead level above 5 μ g/dL. In principle, it is possible that the entire effect could be caused by large effects on development among a few children with very high blood lead levels. However, since the standard deviation in childhood blood lead levels is not extremely large this seems less likely. Still, the exactness of this blood lead level "threshold" remains to be confirmed in future research using individual childhood blood lead levels and individual adult outcomes together with a credible empirical identification strategy.

With these caveats in mind, it is interesting to consider the effects on earnings if early childhood blood lead levels would decrease from 10 μ g/dL to 5 μ g/dL. The Swedish high school graduation premium on earnings at the age of 32 is 17% (Nilsson 2009). By combining this estimate with the estimated effects of lead exposure on high-school graduation rates, the effect of reducing early childhood blood lead levels from 10 to 5 μ g/dL implies that the life-time earnings (among males) would increase by 1.7% (17*0.1). Naturally, this estimate does only capture the part of the lead exposure effects on earnings that goes through high school completion.

In the US alone, 530,000 children aged below five have B-Pb levels above $5\mu g/dL$, and 150,000 out of these are above $10\mu g/dL$ (CDC 2012). Using the above predictions, it is for example possible to calculate the hypothetical annual gains from reducing the B-Pb from 10 to $5\mu g$ among these children. Since general equilibrium effects are most likely not an issue, under the assumption that the earnings effects are directly translatable to the US setting, and given an annual income of USD 30,000, the benefits in terms of increased labor market earnings from reducing the blood lead level in these

children would hence be around USD \$76 million annually after the age of 32 (30,000*0.017*150000). This reflects the effect on the average population of children, but since 60% of all children with a B-Pb above 10 µg/dL are eligible for Medicaid (see Currie, 2009), the expected effects on individual earnings could be larger.

6 Concluding remarks

We document the effects of early exposure to low levels of lead on adult outcomes, by exploiting the phase-out of leaded gasoline, a novel measure of lead exposure, and population-wide administrative individual data covering a wide set of key adulthood outcomes in Sweden. Contrary to the unexplained but "common finding" (in cross-sectional studies) that the marginal effects on cognitive and behavioral effects are larger at low lead levels (EPA 2013), we find that below certain thresholds, the relationship for non-cognitive skills, crime, and high school completion becomes much weaker.

An estimated 535,000 children in the United States have blood lead levels (BLLs) at or above the reference value for blood lead established by CDC in 2012 (5 µg/dL). 150,000 of these children's levels are ≥ 10 µg/dL. Globally, WHO have estimated that more than 40% of the children have blood lead levels above 5 µg/dL. Our results suggest that early lead exposure may have deleterious effects on the academic performance among children with blood lead at least at levels from 3 µg/dL and above, and criminal behavior from 5 µg/dL. The effects sizes are of a clear economic significance. While the exposed cohort is still too young to get a clear estimate of the effects on their labor market outcomes, our back of the envelope calculation suggests that going from 10 to 5 µg/dL implies a 10 percent increase in the probability of high school completion and, taking the non-linear effect into account, a 2.9 percentage point decrease in the probability of being convicted for property crime among males. In terms of earnings, solely via the effects on high school graduation, the same decrease corresponds to an increase by approx. 1.7%. Naturally, if the effects on crime and non-cognitive skills are also taken into account, the effects are likely to be larger.

In the United States, the age of the housing stock together with the neighborhood poverty status constitute a key predictor of elevated blood lead levels in children. Interestingly, we find that boys are more vulnerable to lead exposure, which suggests that early lead exposure could be one contributing explanation for why growing up in disadvantaged areas is especially harmful for boys (Chetty et al. 2016). There is no evidence that early childhood blood lead levels differ by gender and hence, an important task for future research is to provide evidence on whether the larger effects on boys are explained by greater vulnerability, or whether parents, teachers or others differ in their responsive investments to behavioral changes induced by lead exposure between boys and girls.

In terms of mechanisms, non-cognitive skills seem to play a more important role than cognitive skills. This is an important finding since earlier studies have documented that non-cognitive skills are more malleable than cognitive skills. Hence, our study does not only provide causal evidence on the effects of low levels of lead exposure on adult outcomes, but we also provide key insights for effective parental and public remedial investments following early lead exposure. Our results suggest that interventions that specifically target non-cognitive skills are likely to be more successful than those focusing on cognitive skills.

References

- Agency for Toxic Substances and Disease Registry (2007), "Toxicological Profile for Lead"
- Aizer, A., Currie, J., Simon, P. and P. Vivier (2015), "Lead Exposure and Racial Disparities in Test Scores", Working Paper, Brown University
- Andersson, H., Petersson-Grawe, K., Lindqvist, E., Luthman, J., Oskarsson, A. and L. Olson (1997), "Low-level Cadmium Exposure of Lactating Rats Causes Alterations in Brain Serotonin Levels in the Offspring", *Neurotoxicol. Teratol.* 19: 105–115
- Åkerlund, D., Golsteyn, B., Grönqvist, H. and L. Lindahl (2016), "Time Discounting and Criminal Behavior", *Proceedings of the National Academy of Sciences* (in press)
- Almlund, M., Duckworth, A., Heckman, J., and T. Kautz, (2011), "Personality Psychology and Economics", In E.A. Hanushek, S. Machin and L. Woessmann (Eds.), Handbook of the Economics of Education. Amsterdam: Elsevier
- Baltagi, B. H. och D. Li. (2002), "Series estimation of partially linear panel data models with Fixed effects", *Annals of Economic and Finance* 3: 103-116
- Baker M, Gruber, J. and K. Milligan (2008), "Universal Childcare, Maternal Labor Supply, and Family Well-being", *Journal of Political Economy* 116(4): 709–745
- Banks, E.C., Ferretti, L.E. and D.W. Shucar (1997), "Effects of Low Level Lead Exposure on Cognitive Function in Children: A Review of Behavioral, Neuropsychological, and Biological Evidence", *Neurotoxicology*, 18(1): 237–81
- Bellinger D.C. (2004), "Assessing Environmental Neurotoxicant Exposures and Child Neurobehavior: Confounded by Confounding?", *Epidemiology* 15: 383–384
- Billings, S. and K. Schnepel (2015), "Life Unleaded: Effects of Early Interventions for Children Exposed to Lead", Life Course Centre Working Paper No. 2015–18
- Böhlmark A. and M. Lindquist (2006), "Life-Cycle Variations in the Association between Current and Lifetime Income:Replication and Extension for Sweden", *Journal of Labor Economics*, vol. 24, no. 4
- Borghans, L., Golsteyn, B., Heckman, J. and J. Humphries (2011), "Identification Problems in Personality Psychology", *Pers Individ Dif*, 51(3): 315–320

- Borghans, L., Golsteyn, B., Heckman, J. and J. Humphries (2015), "What do Grades and Achievement Tests Measure?", *Pers Individ Dif*, 51(3): 315–320
- Burns, J., Baghurst P., Sawyer, M., McMichael, A. and S. Tong (1999), "Lifetime Lowlevel Exposure to Environmental Lead and Children's Emotional and Behavioral Development at Ages 11–13 years: The Port Pirie Cohort Study", *American Journal of Epidemiology*, 149(8): 740–749
- Case A., Lubotsky, D. and C. Paxson (2002), "Economic Status and Health in Childhood: The Origins of the Gradient", *American Economic Review*, 92(5): 1308– 1334
- Canfield, R., Henderson, C., R. Cory-Slechta, et al. (2003), "Intellectual Impairment in Children with Blood Lead Concentrations below 10 microgram per Deciliter", *New England Journal of Medicine*, 348: 1517–1526
- Cameron, C. and D. Miller (2015), "A Practitioner's Guide to Cluster-Robust Inference", *Journal of Human Resources*, 50: 317–372
- CDC (2010), "Low Level Lead Exposure Harms Children: A Renewed Call of Primary Prevention"
- Cecil, K.M., Brubaker, C.J., Adler, C.M., Dietrich, K.N., Altaye, M., Egelhoff, J.C., et al. (2008), "Decreased Brain Volume in Adults with Childhood Lead Exposure", *PLoS Med* 5:e112.doi:10.1371/journal.pmed.0050112
- Chay, K., and M. Greenstone (2003), "The Impact of Air Pollution on Infant Mortality: Evidence from Geographic Variation in Pollution Shocks Induced by a Recession", *Quarterly Journal of Economics*, 128(4): 1121–1167
- Chetty R., Hendren, N., Lin, F., Majerovitz J. and B. Scuderi (2016), "Childhood Environment and Gender Gaps in Adulthood", *American Economic Review Papers and Proceedings*, 106(5): 282–88
- Clay, K., Troesken, W. and M. Haines (2014), "Lead and Mortality", *Review of Economics and Statistics*, 96: 458–470
- Cunha, F. and J Heckman (2007), "The Technology of Skill Formation", American Economic Review, 97(2): 31–47

- Cunha, F. and J. Heckman (2009), "The Economics and Psychology of Inequality and Human Development", *Journal of the European Economic Association*, 7(2–3): 1–48
- Currie, J. and R. Hyson (1999) "Is the Impact of Health Shocks Cushioned by Socioeconomic Status?: The Case of Birth Weight", *American Economic Review*, 89(2): 245– 250
- Currie, J. and M. Neidell (2005), "Air Pollution and Infant Health: What Can We Learn from California's Recent Experience?", *The Quarterly Journal of Economics*, 120(3): 1003–1030
- Currie, J. and J. Schmieder (2009), "Fetal Exposure to Toxic Releases and Infant Health", *American Economic Association Papers and Proceedings*, 99(2): 177–183
- Currie J., Stabile, M., Manivong, P. and L. Roos (2009), "Child Health and Young Adult Outcomes", *Journal of Human Resources* (forthcoming)
- Danielson, L. (1970), "Gasoline Containing Lead", Swedish Natural Science Research Council., Ecological Research Committee Bulletin 6, 0375–2208
- Davidson P, and D. Anderton (2000), "Demographics of Dumping II: A National Environmental Equity Survey and the Distribution of Hazardous Materials Handlers", *Demography*, 37(4): 461–466
- Dietrich, K., Ris, D., Succop, P. et al. (2001), "Early Exposure to Lead and Juvenile Delinquency", *Neurotoxicol Teratol*, 23(6): 511–518
- Elinder, C-G., L. Friberg, B. Lind, B. Nilsson, M. Svartengren, I. Övermark (1986), "Decreased blood lead levels in residents of Stockholm for the period 1980–1984", *Scand. J. of Work, Environment & Health.* Vol.12 No.2, April.
- Ernhart, C. B. (1995), "Inconsistencies in the Lead-effects Literature Exist and Cannot be Explained by "Effect" Modification". *Neurotoxicology and Teratology*, 17, 227– 233
- Etzel R.A. (2003), "American Academy of Pediatrics Committee on Environmental Health", Pediatric Environmental Health, 2nd ed. Elk Grove Village, IL: American Academy of Pediatrics

- Feigenbaum, J. and C. Muller (2016), "Lead Exposure and Violent Crime in the Early Twentieth Century", *Explorations in Economic History* (forthcoming)
- Ferrie, J., Karen, R. and W. Troesken (2012), "Cognitive Disparities, Lead Plumbing, and Water Chemistry: Prior Exposure to Water-borne Lead and Intelligence Test Scores among World War Two U.S. Army Enlistees", *Economics & Human Biology*, 10(1): 98–111
- Gazze, L. (2016), "Lead Policies, Lead Poisoning, and Government Spending", Mimeo, MIT
- Grönqvist, H., Nilsson, J. P., and P-O. Robling (2015), "Early Childhood Lead Exposure and Criminal Behavior", Working-paper SOFI, Stockholm University
- Grönqvist, H., Niknami, S. and P-O. Robling (2015), "Childhood Exposure to Segregation and Criminal Behavior: Evidence from the "Whole of Sweden Strategy", Working Paper, SOFI Stockholm University
- Hanna-Attisha, M., LaChance, J., Sadler, R.C., and A. Champney Schnepp (2016), "Elevated Blood Lead Levels in Children Associated With the Flint Drinking Water Crisis: A Spatial Analysis of Risk and Public Health Response", *American Journal of Public* Health, 106(2): 283–290
- Hansen, B. (1999), "Threshold Effects in Non-dynamic Panels: Estimation, Testing, and Inference", *Journal of Econometrics*, 93: 345–368
- Harrison R, and J. Yin (2000), "Particulate Matter in the Atmosphere: Which Particle Properties are Important for its Effects on Health?", *Science of the Total Environment*, 249: 85–101
- Hebben, N. (2001), "Low Lead Levels and Neuropsychological Assessment: Let's not be Mislead", Archives of Clinical Neuropsychology, 16 (2001), pp. 353–357
- Heckman, J., Stixrud, J. and S. Urzua (2006), "The Effects of Cognitive and Noncognitive Abilities on Labor Market Outcomes and Social Behavior", *Journal of Labor Economics*, 24(3): 411–482
- Heckman, J., Pinto, R. and P. Savelyev (2013), "Understanding the Mechanisms Through Which an Influential Early Childhood Program Boosted Adult Outcomes", *American Economic Review*, 103(6): 2052–2086

- Hjortenkrans, D., Bergbäck, B. and A. Häggerud, (2007), "Metal Emissions from Brake Linings and Tires: Case Studies of Stockholm, Sweden 1995/1998 and 2005", *Environ. Sci. Technol.*, 41(15): 5224–5230
- IPCS (1992), International Programme on Chemical Safety *Environmental health criteria 134. Cadmium*, Geneva: World Health Organization
- Isen, A., Rossin-Slater, M. and R. Walker (forthcoming), "Every Breath You Take | Every Dollar You'll Make: The Long-Term Consequences of the Clean Air Act of 1970", *Journal of Political Economy*
- Jans, J., Nilsson, P. and P. Johansson (2014), "Economic Status, Air Quality, and Child Health: Evidence from Inversion Episodes", IZA Discussion paper 7929
- Johansson, C., Norman, M., and L. Burman (2009), "Road Traffic Emission Factors for Heavy Metals", Atmospheric Environment, 43, 4681–4688
- Kaufman, A. S. (2001a), "Do Low Levels of Lead Produce IQ Loss in Children? A Careful Examination of the Literature", *Archives of Clinical Neuropsychology*, 16(4): 303–341
- Kaufman, A. S. (2001b) "How Dangerous are Low (not moderate or high) Doses of Lead for Children's Intellectual Development?", Archives of Clinical Neuropsychology, 16(4): 403–431
- Landrigan, P. J., Gehlbach, S. H., Rosenblum, B. F., Shoults, J. M., Candelaria, R. M., Barthel, W. F., Liddle, J. A., Smrek, A. L., Staehling, N. W., and J. F. Sanders (1975), "Epidemic Lead Absorption Near an Ore Smelter: The Role of Particulate Lead", New England Journal of Medicine, 292(3): 123–129
- Lanphear B., Dietrich, K., Auinger, P. and C. Cox. (2000), "Cognitive deficits associated with blood lead concentrations <10 microg/dL in US children and adolescents", *PublicHealth Rep.* 115(6): 521–529
- Lidsky T.I. and J.S. Schneider (2003), "Lead Neurotoxicity in Children: Basic Mechanisms and Clinical Correlates", *Brain*, 126: 5–19
- Lindqvist, E. and R. Vestman (2011), "The Labor Market Returns to Cognitive and Noncognitive Ability: Evidence from the Swedish Enlistment", American Economic Journal: Applied Economics, 3(1): 101–128

- Liu, J., Liu, X., Wang, W., et al. (2014), "Blood Lead Concentrations and Children's Behavioral and Emotional Problems: A Cohort Study", *JAMA Pediatr.*, 168(8): 737–745
- Lüchinger, S. (2009), "Air Pollution and Infant Mortality: A Natural Experiment from Power Plant Desulfurization", mimeo LSE
- MOENR (1994), *The Swedish Experience Taxes and Charges in Environmental Policy*. Ministry of the Environment and Natural Resources. Stockholm, Sweden
- Moon C.S., Paik, J.M., Choi, C.S., Kim, D.H. and M. Ikeda (2003), "Lead and Cadmium Levels in Daily Foods, Blood and Urine in Children and Their Mothers in Korea", *Int Arch Occup Environ Health*, 76: 282–´–288
- Needleman, H., Gunnoe, C., Leviton, A., Reed, R., Pesesie, H., Maher, C. and P. Barrett (1979), "Lead Levels and Children's Psychologic Performance." *New England Journal of Medicine*, 301(3): 689–695
- Needleman, H., Riess, J., Tobin, M. et al. (1996), "Bone Lead Levels and Delinquent Behavior", *Journal of the American Medical Association*, 275(5): 363–369
- Needleman, H, and D. Bellinger (2001), "Studies of Lead Exposure and the Developing Central Nervous System: A Reply to Kaufman", Archives of Clinical Neuropsychology, 16(2001): 359–374
- Needleman, H. (2004), "Lead Poisoning", Annual Review of Medicine, 55: 209-222
- Neidell, M. (2004), "Air Pollution, Health, and Socio-economic Status: The Effect of Outdoor Air Quality on Childhood Asthma", *Journal of Health Economics*, <u>23(6)</u>: 1209–1236
- Nilsson, J. P. (forthcoming), "Alcohol Availability, Prenatal Conditions, and Long-term Economic Outcomes", *Journal of Political Economy*
- Nilsson, J. P. (2009), "The Long-term Effects of Early Childhood Lead Exposure: Evidence from the Phase-out of Leaded Gasoline", Chapter included in doctoral thesis, Department of Economics, Uppsala University, December 2009
- Nilsson, J P, S. Skerfving, E. Stroh, U. Strömberg (2009), "The Relationship Between Ambient Air Pollution and Pollution Exposure in Children", mimeo, Uppsala University

- Nordell, O. (2007), "Tungmetaller i mossor i Landskrona kommun 1983, 1995 och 2006", Rapport 2007:3, Miljöförvaltningen Landskrona
- OECD (2014), Education at a Glance 2014: OECD Indicators, OECD Publishing.
- Olsson, I., Eriksson, J., Öborn, I., Skerfving, S., and A. Oskarsson (2005), "Cadmium in Food Production Systems: A Health Risk for Sensitive Population Groups", *Ambio*, 34: 344–351
- Onianwa P.C. (2001), "Monitoring Atmospheric Metal Pollution: A Review of the Use of Mosses as Indicators", *Environmental Monitoring and Assessment*, 71: 13–50
- Petersson-Grawe, K., Teiling-Gardlund, A., Jalkesten, E. and A. Oskarsson (2004), "Increased Spontaneous Motor Activity in Offspring after Maternal Cadmium Exposure during Lactation", *Environ. Toxicol. Pharmacol*, 17: 35–43
- Rehnberg, I., Brännvall, M., Bindler, R. and O. Emteryd (2000), "Atmospheric Lead Pollution History during Four Millennia (2000 BC to 2000 AD) in Sweden", *Ambio*, 29(3)
- Reyes, J. (2007), "Environmental Policy as Social Policy? The Impact of Childhood Lead Exposure on Crime," *The B.E. Journal of Economic Analysis & Policy*, 7(1) (Contributions), Article 51
- Reyes, J. (2011), "Childhood Lead and Academic Performance in Massachusetts", *New England Public Policy Center Working Paper*, 11(3)
- Reyes, J. (2014), "Lead Exposure and Behavior: Effects on Antisocial and Risky Behavior among Children and Adolescents", NBER Working Paper No. 20366
- Rühling, Å., and G. Tyler (1968), "An Ecological Approach to the Lead Problem", *Botaniska Notiser*, 121: 321–342
- Rühling, Å. and G. Tyler (1969), "Ecology of Heavy Metals A Regional and Historical Study", *Botaniska Notiser*, 122: 248–259
- Skerfving S, and I. Bergdahl "Lead", In: Nordberg G.F., Fowler B.A., Nordberg M., Friberg L.T. Handbook on the Toxicology of Metals, Academic Press, Elsevier, 599– 643

- Skerfving S., Löfmark, L., Lundh, T., Mikoczy, Z. and U. Strömberg (2015), "Late Effects of Low Blood Lead Concentrations in Children on School Performance and Cognitive Functions", *NeuroToxicology*, 49: 114–120
- Strömberg, U., Schütz, A. and S. Skerfving (1995), "Substantial Decrease of Blood Lead in Swedish Children, 1978–94, Associated with Petrol Lead", Occup Environ Med, 52: 764–769
- Strömberg U., Lundh, T., Schütz, A. et al. (2003), "Yearly Measurements of Blood Lead in Swedish Children Since 1978: An Update Focusing on the Petrol Lead Free Period 1995–2001", Occup Environ Med, 60: 370–372
- Thomas, V., Socolow, R., Fanelli, J. and T. Spiro (1999), "Effects of Reducing Lead in Gasoline: An Analysis of the International Experience", *Environ. Sci. Technol*, 33(22): 3942–3948
- Troesken, W. (2008), "Lead Water Pipes and Infant Mortality at the Turn of the Twentieth Century", *Journal of Human Resources*, 43: 553–575
- Weiss, B. and J. Elsner (1996), "The Intersection of Risk Assessment and Neurobehavioral Toxicity", *Environ. Health Perspect*, 104(Suppl 2): 173–177
- World Health Organization (1972), "Sixteenth Report of the Joint FAO/WHO Expert Committee on Food Additives", WHO Technical Report Series no. 505. Geneva: WHO
- World Health Organization (2016), "Lead Poisoning and Health", Fact sheet, updated August 2016, *http://www.who.int/mediacentre/factsheets/fs379/en/*
- Wright, J., Kim, P., Dietrich, N., Douglas M., et al. (2008), "Association of Prenatal and Childhood Blood Lead Concentrations with Criminal Arrests in Early Adulthood", *PLoS Med*, 5(5), e101



Appendix A.I.: Figures and tables

Figure A1. Neighborhood of birth lead exposure levels in 1975



Figure A2. Neighborhood of birth lead exposure levels in 1985



Figure A3. Kernel density distributions of moss lead levels across the neighborhoods



Figure A4. Distribution of changes in neighborhood lead levels



Figure A5. The association between changes in municipality moss lead concentrations and changes in (log) population density



Figure A6. The association between changes in municipality moss lead concentrations and changes in municipality employment rate





Notes: The graphs present estimates along with a 95 % confidence interval from semi-parametric fixed effects models (see Baltagi and Li 2002). The dependent variable in the top graph is grade 9 GPA and in the bottom graph P(High school completion). The sample consists of males born in 1972–1974, 1977–1979 and 1982–1984. Childhood lead exposure is measured as an average over ages 1–3. See notes of Table 1 for full set of controls. The confidence interval accounts for clustering at the commuting zone level.





Notes: The graphs present Hansen's (1999) likelihood ratio test to locate the true threshold. The top graph presents results for compulsory school GPA and the bottom graph presents results for P(High school completion). The threshold is the assumed threshold at which the sum of squared residuals in the piecewise linear regression is minimized. The 5% critical value of the likelihood ratio is 7.35.



Figure A9. Hansen's LR statistic for crime outcomes. (MALES)

Notes: The graphs present Hansen's (1999) likelihood ratio test to locate the true threshold. The top graph presents results for P(Ever convicted), the middle for P(Violent crime) and the bottom for P(Property crime). The dashed horizontal line is the critical value for the no rejection region (i.e. it is not possible to reject that the true threshold is located at the locations where the LR takes a value lower than the critical value). The threshold is the assumed threshold at which the sum of squared residuals in the piecewise linear regression is minimized. Note that for Violent crime and Property crime we identify thresholds; however, for the ever convicted outcome, there is no location that is significant. The 5% critical value of the likelihood ratio is 7.35.



Figure A10. Maximum gasoline lead sales in metric tons per year

Note: The figure provides the maximum total lead sales over the years 1976–1984 using yearly data on deliveries of motor fuel (regular, medium, and premium grade gasoline) to final consumers. We multiply the delivered volumes by the fuel type specific maximum allowed lead content (for regular and premium gasoline, see Figure 1).





Note: These figures show the point-estimates, point-wise 95% confindence bands from the estimation of the cohort-specific $\gamma's$ from equation (5) on the male sample together with the F-stats and p-values for a test of the equality of the γ coefficients before and after the reform (i.e. difference-in-differences for 1976–1978 vs.1979–1984 cohort coefficients). The top (bottom) left figure provides the estimates for 9th grade GPA among children born in neighborhoods with a lead level of above (below) 30mg/kg just before the reform. The top (bottom) right figure provides the estimates for high-school completion among children born in neighborhoods with a lead level of above (below) 30mg/kg just before the reform. The point estimates (and standard errors) provided in these four figures constitute the basis for the top panel of Figure 4. See text and notes to Figure 4 for further details.



Figure A12. Violent and property crime above and below the human capital threshold (MALES)

Note: These figures show the point-estimates, point-wise 95% confindence bands from the estimation of the cohort-specific $\gamma's$ from equation (5) on the male sample together with the F-stats and p-values for a test of the equality of the γ coefficients before and after the reform (i.e. difference-in-differences for 1976–1978 vs.1979–1984 cohort coefficients). The top (bottom) left figure provides the estimates Violent Crime among children born in neighborhoods with a lead level of above (below) 30mg/kg just before the reform. The top (bottom) right figure provides the estimates for Property Crime among children born in neighborhoods with a lead level of above (below) 30mg/kg just before the reform. The point estimates (and standard errors) provided in these four figures constitute the basis for the Bottom Panel of Figure 4. See text and notes to Figure 4 for further details.



Figure A13. Any crime conviction in neighborhoods above and below the human capital threshold (MALES)

*Note:*See Figure A12 for details.



Figure A14. Human capital above and below the threshold (FEMALES)

Note: These figures show the point-estimates, point-wise 95% confindence bands from the estimation of the cohort-specific γ 's from equation (5) on the female sample together with the F-stats and p-values for a test of the equality of the γ coefficients before and after the reform (i.e. difference-in-differences for 1976–78 vs.1979–1984 cohort coefficients). The top (bottom) left figure provides the estimates for 9th grade GPA among children born in neighborhoods with a lead level of above (below) 30mg/kg just before the reform. The top (bottom) right figure provides the estimates for high-school completion among children born in neighborhoods with a lead level of above (below) 30mg/kg just before the reform.

	Definitions	Mean	Std. dev.
GPA	Compulsory school grade point average (percentile ranked) [typically	50.09	28.8
Cognitive Skills	Cognitive test score at military enlistment (percentile ranked) [age 18; boys only]	50.0	28.5
Non-Cognitive skills	Non-cognitive test score at military enlistment (percentile ranked) [age 18; boys only]		
High School	=1 if completed high school, 0 otherwise	.89	.31
Ever convicted	=1 if ever convicted in a criminal trial up to age 24, 0 otherwise	.164	.370
Property crime	=1 if ever convicted for a property crime (penal code: BRB Chapter 8) up to age 24, 0 otherwise	.070	.155
Violent crime	=1 if ever convicted for a violent crime (penal code: BRB Chapter 3) up to age 24, 0 otherwise	.024	.256

Table A1. Descriptive statistics for outcomes used in the study

Table A2. Parents' predetermined characteristics and their children's lead exposure

Table Tall Talente	predeterm						
PANEL A:	Parents completed college	Mother completed college	Father completed college	Parents' earnings 2 years before birth	Father's earnings 2 years before birth	Mother's earnings 2 years before birth	Age of mother at birth
Lead exposure (mg/Kg) Neighborhood fixed effects?	0005 (.0004) No	0004 (.0004) No	0003 (.0004) No	.0004 (.0003) No	.0004 (.0003) No	0001 (.0006) No	0004 (.0029) No
PANEL B:	Parents completed college	Mother completed college	Father completed college	Parents' earnings 2 years before birth	Father's earnings 2 years before birth	Mother's earnings 2 years before birth	Age of mother at birth
Lead exposure (mg/Kg) Neighborhood fixed effects?	.0001 (.0001) Yes	.0001 (.0001) Yes	0000 (.0001) Yes	.0003 (.0003) Yes	.0003 (.0002) Yes	.0003 (.0003) Yes	0014 (.0028) Yes

Notes: Standard errors (in parenthesis) are clustered at the commuting zone level. All regressions control for cohort of birth fixed effects.

Appendix A.II.: Sibling fixed effects

Table A3 shows results from sibling fixed effects estimates. Panel A provides OLS estimates for the sibling sample. Due to the gender specific effects and that we can only measure lead exposure in specific years, the sampling of siblings is somewhat more complex than in a standard siblings fixed effects model. In particular, we restrict the sample to mothers with at least two but at most three males born during the observation windows (cohorts 72–74, 77–79, 82–84). In addition, to exploit the policy driven variation rather than (potentially endogenous) changes in exposure from families that move between births, we require that at least one of the brothers must be born in a different lead measurement period (remember that we measure average lead over the three preceding years in 1975, 1980 and 1985).

For a comparison with the full sample, Panel A of Table A3 first provides the OLS estimates using this restricted sibling sample. Overall, the OLS estimate of the full sample is reasonably similar to the sibling sample estimates for males. Compared to the sibling sample, the full sample GPA and high school completion estimates are slightly higher, while the estimates for ever convicted and violent crime are higher, and similar for property crime. Turning to the siblings fixed effects estimates in Panel B, we see that as compared to the OLS estimates, the precision is first of all slightly worse, but the point estimates are not systematically different from the OLS estimates, which is reassuring. Some estimates are larger while others are smaller than the OLS counterparts, and none are statistically distinguishable from each other. Overall, we take this as supporting evidence that the estimates from the baseline empirical strategy for the full sample of children are valid.

Table A3. Sibling fixed effects estimates

	Grade 9 GPA (pct. rank)	P(High school completion)	P(Ever convicted)	P(Property crime)	P(Violent crime)
	(1)	(2)	(3)	(4)	(5)
	Panel A: OLS or	Sibling Sample			
Early lead exposure	2112** (.1062)	0044* (.0025)	.0079 ^{***} (.0023)	.0016 (.0023)	.0028 [*] (.0016)
	Panel B: Sibling	fixed effects			
Early lead exposure	1410 (.1810)	0050 (.0035)	.0065 ^{**} (.0031)	.0024 (.0023)	.0015 (.0030)
Neighborhood FEs	Yes	Yes	Yes	Yes	Yes
Year of birth FEs	Yes	Yes	Yes	Yes	Yes

Notes: The table presents the coefficient on early lead exposure from separate regressions. Panel B includes maternal fixed effects for individuals with up to three siblings. Coefficients and standard errors are scaled by a factor 10. Crime (in columns 3-5) is defined as having been convicted at least once up to age 24. The sample consists of male children born in 1972–1974, 1977–1979 and 1982–1984. Childhood lead exposure is measured as an average over ages 1-3. The baseline model includes controls for each parent's highest completed level of education (4 levels), total family earnings, maternal age at birth, birth order, family size, municipality population size, municipality employment rate, Cadmium, Zinc and Copper. Cluster robust standard errors (at the commuting zone level (74 cells)) are shown in parenthesis. *** = significant at 1 %, * ** = significant at 5 %, * = significant at 10 %.

Appendix B: The relationship between lead exposure and blood lead among children

The relationship between blood lead and moss lead

This section briefly reviews the results from Nilsson, Skerfving, Stroh and Strömberg (2009) that provide the estimated elasticity between lead levels in moss and blood-lead levels in children. The interested reader is referred to that study for further details on the data and execution.

The moss samples that Nilsson et al. (2009) use were collected at 55 sites in the municipality of Landskrona in 1983, 1995 and 2006, following the same principles as in the national bio-monitoring program. These data were then matched to the (venous) blood lead measurements from 396 children aged between 7 and 10, collected by Strömberg et al. (2003) in the year prior to that during which the mosses were sampled. Using the coordinates of the children's home address, each child is assigned an inverse distance weighted lead exposure level using the 10 nearest moss sampling sites.⁴² The raw correlation between this lead exposure measure and children's blood lead level is 0.60, which compares very well with findings in previous studies linking ambient air pollution to actual population exposure.

Table A1 reports the estimated elasticity between lead in moss and children's blood lead levels using six different versions of the following specification,

$$\ln(blood_lead)_{it} = \alpha + \gamma \ln(exposure)_{it} + X'\beta + \theta_c + \theta_t + \varepsilon_{it}$$
(A1)

In the first column of Table B1., the elasticity between B-Pb and M-Pb using the full sample is shown without any additional control variables added to the model. The estimated coefficient suggests that for a 10% increase in the lead level in moss, the blood lead level increases by on average 3%. In columns (2)–(4), individual characteristics, fixed community, year of sampling and finally year*community fixed effects are stepwise introduced. Adding year and community fixed effects is the only control which influences the estimated elasticity.

 $^{^{42}}$ Following Currie and Neidell (2003), in order to assess the accuracy of the air pollution measure, Nilsson et al. compare the actual level of pollution at each moss sample site with the level of pollution that they would have assigned using the implemented method (i.e. using the five closest measuring sites), if the actual moss sample was not, in fact, available. The correlation of the actual and estimated level is high for Pb (*r*=.73), suggesting that it is an accurate measure for the air pollution exposure for the children's home address. Also note that as long as the measurement errors in assigned and actual exposure are not systematic, the relationship between the children's blood-lead levels and our air pollution measure will be biased towards zero.

Dependent variable:	ln (BloodPb)	ln (BloodPb)	In (BloodPb)	In (BloodPb)	ln (BloodPb)	In (BloodPb)
Specification:	(1)	(2)	(3)	(4)	(5)	(6)
Time period:	All	All	All	All	Pre-Ban (<1995)	Pre-Ban (<1995)
In(Moss-Pb)	.2710*** (.0163)	.337*** (.0203)	.1834** (.0914)	.2111** (.0973)	.453*** (.1388)	.463*** (.1183)
Individual controls	no	yes	yes	yes	yes	Only
Neighb. fixed effects	no	no	yes	yes	yes	yes
Year fixed effects	no	no	yes	yes	yes	yes
Year* Neighborhood fixed effects	no	no	no	yes	yes	yes
R-squared	0.35	0.49	0.60	0.61	0.58	0.57
No of children	396	396	396	396	242	242

Table B1. Baseline model for lead

Note: The table reports regression results from OLS estimations of equation (B1). The blood lead is a measure in μ g/dL blood. The lead exposure is mg/kg of moss. The full set of individual controls are gender, whether the child's practicing any lead exposing hobbies, ln(hemoglobin) level and school fixed effects. Standard errors are reported in parenthesis and are robust with respect to heteroscedasticity. */***/*** reflects significance at the 10/5/1 percent levels, respectively. Column (1) provides raw correlation, (2) adds individuals controls, (3) neighborhood and year fixed effects, (4) neighborhood by year effects, (5) restricts the sample to the pre-ban period, (6) drops the more esoteric control variables (i.e. hemoglobin level, and lead hobbies, and school FE) which are typically not available in most datasets. Nilsson, Skerfving, Stroh and Strömberg (2009)

The first four columns report the estimated elasticity using the full sample. However, in 1995, lead in gasoline was banned. Hence, as the relative contribution of air lead for total body burden decreases, the predictive power of the moss samples is likely to decrease as mosses only take up lead from the air. This is mirrored in Figure 1, which shows that while the lead in moss continued to decrease throughout the observation period, the children's blood-lead levels leveled off at around 2 μ g/dL after the ban on leaded gasoline. This pattern is clearly in line with a shift away from air-borne sources as the major source of lead exposure in children after lead was phased out of gasoline. In column (6), we test this notion by estimating the full model only on the two cohorts sampled before the ban on leaded gasoline, i.e. in 1984 and 1994. When using this restricted sample, the estimated elasticity increases to 0.45 while the standard errors increase marginally. This result suggests that the relative importance of air-lead exposure was indeed stronger in the period prior to the ban than in the period after, as expected. The pre-ban period is also the period focused on in this paper. The final result also provides suggestive evidence on the validity of using mosses as monitors of air pollution.

Initial blood lead levels

The earliest systematic blood lead levels sampled among children in Sweden were conducted in 1978 by Strömberg et al. (1995). At that time, the blood lead level was just above 6 μ g/dL on average in two locations in southern Sweden (where the air lead levels were among the highest in Sweden at the time). Needleman reports blood lead levels in the US in the same year of on average around 14 μ g/dL. Before this point in time, we use the estimates of previous work by Reyes (2007) in order to assess the children's lead levels. She finds that an increase of 1 gram lead per gallon of gasoline increases the blood lead level by approximately 3.3 μ g/dL in the general population. Taking her baseline specification literally and combining it with the data in Table 1 and the estimated intercept from Strömberg et al. (1995) B-Pb = 1.67 + 3.325 * grams/gallon) suggests that at the year of birth of the first cohort born in 1972–74, the average blood lead level would on average have been 6.7 μ g/dL.

A second way of assessing the initial blood lead levels is to use the model developed in Nilsson et al. (2009) and use the average moss lead level in 1975 to predict the blood lead levels in primary school children at that time. This approach provides a predicted initial blood lead level of around 3μ g/dL in primary school children. After adjusting the blood lead moss lead elasticity using the age-specific blood lead blood gasoline elasticity estimated by Reyes (2007) (30% higher for children aged 0–6 than for children aged 6–12) and under the additional assumption that the additive separable specification used in the estimation holds for both populations, the relevant blood lead level for children aged 0–6 would, on average, correspond to about 5μ g/dL.

A third way of estimating the initial blood lead level is to use the estimates in Strömberg et al. Based on repeated individual blood lead measurements, they find that individual blood lead levels in primary school children on average decrease by around 6% per year. Given the average blood lead level in 1978, this would imply that the average blood lead levels among 1–4 year olds in 1972 (since the samples are taken from children aged 7–10 in 1978) would, on average, be 8.5μ g/dL. Since these samples were taken in a region with high lead exposure (based on moss lead values), it seems reasonable to assume that this level represents an upper bound of blood lead levels in the general population of children in these cohorts.

All three approaches provide estimates suggesting that the initial blood lead levels in the sample period were on average below $10\mu g/dL$ for the cohorts born between 1972

and 1974. Using the blood lead level standard deviation (1.5 μ g/dL) of school children (age 10) in Strömberg et. al suggests that 97.5% of the first cohort of children in our sample had blood lead levels below 8 μ g/dL, 9.7 μ g/dL, or 11.5 μ g/dL depending on the preferred estimate of the early childhood mean blood-lead level of the children born in 1972–1974.

How does lead exposure affect development?

Although lead exposure may have direct health consequences such as anemia, kidney failure or ultimately death, these effects typically show up only at high exposure levels. The possibility that lower doses may also be harmful for brain development was not recognized until recently. Because lead mimics the actions of calcium, it can pass the blood-brain barrier and thereby affect the central nervous system where it affects the brain's prefrontal cerebral cortex, hippocampus and cerebellum (Finkelstein, Markowitz and Rosen 1998). Cellular and animal studies have confirmed the hypothesis that lead exposure during critical stages of development disrupts the formation of neuron networks and the process of neurotransmission in ways that increase the risk for these types of behavioral problems (Weiss and Elsner 1996). Infants, toddlers, and the developing fetus are at the greatest risk of toxicity from low-level exposure because of a higher absorption rate into the blood and, also conditional on absorption, lead is more likely to affect the developing nervous system than the mature brain (Lidsky and Schneider 2003). Blood-lead levels follow an inverted v-shaped pattern between the ages 6–60 months, reaching its peak at the age of 24 months due to the intense handmouth activity at these ages (Needleman and Bellinger 2001; Canfield et al. 2003).

All these physiological impacts may hurt child development and thereby risk causing permanent damage to children's long-run human capital formation. The specific pathways are still unclear, however. Several epidemiological studies have documented strong correlations between lead and cognitive ability (Lamphear et al., 2005). The effect of lead on children's behavior is less understood and few studies have been done at levels of exposure <10 μ g/dL (Liu et al. 2014). This is important since research has shown that non-cognitive skills are strongly linked to human capital accumulation (e.g. Heckman et al. 2006).

Appendix C: Effects on earnings

	In (Earnings)	Earnings	Standarized Earnings	
	Panel A: Girls			
Early lead exposure	.00036 (.00022)	.5202 (.3888)	.00019 (.00014)	
Sample mean	7.54 Panel B: Boys	2303.68	.824	
Early lead exposure	0.000001 (.0003)	0895 (.8430)	00003 (.0003)	
Sample mean	7.989	3219.495	1.152	
Neighborhood FEs	Yes	Yes	Yes	
Year of birth FEs	Yes	Yes	Yes	

Table C1. The effect of earnings in 2013 (ages 29-40)

Notes: The table presents the coefficient on early lead exposure from separate OLS regressions on ln earnings, Earnings (including zeros), and Standardized Earnings (Earnings/(mean Earnings)) in 2013. Coefficients reflect the effect of increasing lead by 1 unit. The sample consists of children born in 1972–1974, 1977–1979 and 1982–1984. Childhood lead exposure is measured as an average over ages 1–3. Cluster robust standard errors (at the commuting zone level (74 cells)) are shown in parenthesis. *** = significant at 1 %, * ** = significant at 5 %, * = significant at 10 %. See Table 1 for full set of controls.



Figure C1. Semi parametric estimates on males and females In(earnings in 2013) (ages 29–40)

Note: See Figure 2 for details of the estimation procedure.



Figure C2. The estimated impact of the 1980/81 reform on earnings in 2013 (MALES)

Notes: These figures show the point-estimates, point-wise 95% confindence bands from the estimation of the cohort-specific $\gamma's$ from equation (5) on the male sample together with the F-stats and p-values for a test of the equality of the γ coefficients before and after the reform (i.e. difference-in-differences for 1976–78 vs.1979–1984 cohort coefficients). The top/middle/bottom left figure provides the estimates for Earnings/Standardized Earnings/In(Earnings) among children born in neighborhoods with a lead level of Below 30mg/kg just before the reform. Theright figure provides the same estimates for children born in neighborhoods with a lead level of above 30mg/kg just before the reform.



Figure C3. The estimated impact of the 1980/81 reform on earnings in 2013 (FEMALES)

Notes: These figures show the point-estimates, point-wise 95% confindence bands from the estimation of the cohort-specific $\gamma's$ from equation (5) on the female sample together with the *F*-stats and *p*-values for a test of the equality of the γ coefficients before and after the reform (i.e. difference-in-differences for 1976-78 vs.1979–1984 cohort coefficients). The top/middle/bottom left figure provides the estimates for Earnings/Standardized Earnings/In(Earnings) among children born in neighborhoods with a lead level of below 30mg/kg just before the reform. The right figure provides the same estimates for children born in neighborhoods with a lead level of above 30mg/kg just before the reform.